The Macroeconomic Impact of Financial and Uncertainty Shocks

Dario Caldara*

Cristina Fuentes-Albero[†]

ero[†] Simon Gilchrist[‡]

Egon Zakrajšek[§]

September 18, 2014

Abstract

The extraordinary events surrounding the "Great Recession" have cast a considerable doubt on the traditional sources of macroeconomic disturbances. In their place, economists have singled out financial and uncertainty shocks as important drivers of economic fluctuations. Distinguishing between these two types of shocks, however, is difficult because increases in uncertainty are frequently associated with a widening of credit spreads, an indication of a tightening in financial conditions. This paper uses the penalty function approach to jointly identify shocks behind changes in financial conditions and economic uncertainty and to trace out the impact of these two types of shocks on the economy. Importantly, the identifying assumptions do not rule out a contemporaneous response of financial conditions to uncertainty shocks or vice versa. The results indicate that (1) financial shocks have a significant adverse effect on economic outcomes and that such shocks were an important source of cyclical fluctuations since the mid-1980; (2) uncertainty shocks have a significant macroeconomic impact in situations where they elicit a tightening of financial conditions; and (3) the rise in uncertainty in response to a financial shock suggests that swings in uncertainty are influenced importantly by changes in financial conditions. These findings are robust across a wide range of uncertainty indicators used in the literature.

JEL CLASSIFICATION: E32, E37, E44

KEYWORDS: time-varying volatility; credit spreads; structural vector autoregression, optimizationbased identification

We thank seminar and conference participants at the Federal Reserve Board; Georgetown University; the Federal Reserve Bank of Philadelphia; Bank of Portugal; Bank of Spain; the Swiss National Bank; the Inter-American Development Bank; the Guanghua School of Management, Tsinghua University; the 2013 SED Annual Meetings; the 2013 Workshop on Macroeconomics, Financial Frictions, and Asset Prices at the University of Pavia; the 9th Dynare Conference; the Summer 2013 Meeting of the North American Econometric Society; the Spring 2014 Midwest Macro Meeting; the 20th Conference in Computing in Economics & Finance; and the 37th Symposium of the Spanish Economic Association. Shaily Patel and Rebecca Zhang provided outstanding research assistance. All errors and omissions are our own responsibility. The views expressed in this paper are solely the responsibility of the authors and should not be interpreted as reflecting the views of the Board of Governors of the Federal Reserve System or of anyone else associated with the Federal Reserve System.

^{*}Federal Reserve Board of Governors. Email: dario.caldara@frb.gov

[†]FEDERAL RESERVE BOARD OF GOVERNORS. Email: cristina.fuentes-albero@frb.gov

[‡]BOSTON UNIVERSITY AND NBER. Email: sgilchri@bu.edu

[§]FEDERAL RESERVE BOARD OF GOVERNORS. Email: egon.zakrajsek@frb.gov

1 Introduction

The acute turmoil that swept through global financial markets during the 2008–09 financial crisis and the depth and duration of the associated economic downturn, both in the United States and abroad, have cast considerable doubt on the traditional sources of business cycle fluctuations. In response, a flurry of theoretical and empirical research aimed at understanding these extraordinary events has pointed to financial and uncertainty shocks—or their toxic combination—as alternative drivers of economic fluctuations (see Bloom, 2009; Bloom et al., 2012; Arellano et al., 2012; Christiano et al., 2014; Gilchrist et al., 2014b). Empirically distinguishing between these two types of shocks, however, is difficult because increases in financial market volatility—a widely used proxy for economic uncertainty—are frequently associated with significant increases in credit spreads, a timely indicator of disruptions in financial markets.¹

A stark illustration of this empirical challenge is depicted in Figure 1, which shows the relationship between the daily change in the option-implied volatility on the S&P 500 stock futures index (the VIX) and the daily change in the speculative-grade CDX index during the "Great Recession."² Clearly evident is the fact that episodes of acute financial distress are associated with significant spikes in asset price volatility. Indeed, in their comprehensive empirical anatomy of the Great Recession, Stock and Watson (2012) explicitly single out the high (positive) correlation between credit spreads and proxies for economic uncertainty and conclude that "[T]hese two sets of instruments do not seem to be identifying distinct shocks."

The identification of financial and uncertainty shocks within a structural vector autoregressive (SVAR) framework—the workhorse of empirical macroeconomics—is complicated by the fact that indicators of financial distress such as credit spreads and volatility-based uncertainty proxies are both "fast-moving" variables. As a result, it is difficult to impose plausible zero contemporaneous restrictions to identify these two types of disturbances. It it also difficult to impose sign restrictions on the impulse response functions in order to achieve an economically plausible identification because financial and uncertainty shocks have theoretically the same qualitative effects on both prices and quantities in most instances.

¹The fact that the level of credit spreads is highly informative about the tightness of financial conditions in the economy and thus the implied degree of departure from the Modigliani–Miller paradigm of frictionless financial markets is supported by the rapidly growing empirical literature showing that corporate bond credit spreads form the most informative and reliable class of financial indicators for future economic activity and that unanticipated increases in credit spreads have large and persistent adverse effects on the macroeconomy (see Gertler and Lown, 1999; Mueller, 2009; Gilchrist et al., 2009; Gilchrist and Zakrajšek, 2012; Bleaney et al., 2012; Boivin et al., 2013; Faust et al., 2013; Gilchrist and Mojon, 2014).

²The VIX index is a commonly used proxy for economic uncertainty (see Bloom, 2009). The speculative-grade CDX index is a tradable credit derivative index used widely by investors for hedging of and investing in corporate credit risk. Buying and selling of the credit derivative index is comparable to buying and selling portfolios of corporate cash instruments: By buying the index, the investor takes on the credit exposure—is exposed to defaults—a position similar to that of buying a portfolio of bonds; by selling the index, the credit default swap (CDS) contracts on firms that have a "junk" rating from either Moody's or Standard & Poor's. The component firms must have highly liquid single-name CDS trading in their name, and the composition of both indexes, which is determined by a dealer poll, is representative of the U.S. corporate sector.



Figure 1: Financial Market Volatility and Credit Risk During the Great Recession

NOTE: Sample period: daily data from 12/01/2007 to 06/30/2009. The scatter plot depicts the relationship between the daily change in the option-implied volatility on the S&P 500 stock futures index (VIX) and the daily change in the 5-year (on-the-run) speculative-grade CDX index. The period 09/01/2008 to 08/05/2009 marks the acute phase of the crisis, which reached a critical stage in early September 2008, when an evaporation of liquidity in the global credit markets threatened the solvency of several major financial institutions. The end date of the acute phase follows the release of the results from the Supervisory Capital Assessment Program (the so-called bank stress test) at 5 p.m. EST on May 7, 2009.

In this paper, we use the penalty function approach developed initially by Faust (1998) and Uhlig (2005) to identify shocks behind changes in financial conditions and economic uncertainty and to trace out the impact of these two types of disturbances on the macroeconomy. Within our VAR framework, the two structural innovations are identified using a criterion that each shock should maximize the *impulse response* of its respective target variable over a pre-specified horizon. This quantitative criterion allows us to distinguish between shocks that have otherwise very similar qualitative effects on the macroeconomy.

Because this approach requires a sequential identification of shocks, we implement the penalty function criterion in two steps. In the baseline identification, we first search for an innovation that maximizes the response of the financial stress variable over a given horizon—this optimization step identifies what we call a "financial shock." In the second step, we search for an innovation that maximizes the response of the uncertainty proxy over the same horizon and that is orthogonal to the financial shock identified in the first step—we call this shock an "uncertainty shock."

We also consider an alternative identification scheme that reverts the ordering of the two penalty function steps to identify these two types of disturbances. Both of our identification strategy effectively assume that the unanticipated changes in financial conditions are due primarily to financial shocks, whereas fluctuations in economic uncertainty are driven mainly by uncertainty shocks. Importantly, neither identification schemes rules out a contemporaneous response of financial conditions to uncertainty shocks or vice versa. It is also worth emphasizing that our approach differs from the pure sign restriction identification schemes in that it identifies a single structural VAR specification, rather than a set of models (see Fry and Pagan, 2011; Caldara and Kamps, 2012; Arias et al., 2013).

In implementing this approach, we consider a standard monetary VAR, augmented with an indicator of financial conditions and a proxy for macroeconomic uncertainty. To measure strains in financial markets, we use the excess bond premium (EBP), an indicator of the effective "risk-bearing capacity" of the financial intermediary sector, developed recently by Gilchrist and Zakrajšek (2012). In the benchmark VAR specification, we use the realized stock market volatility as a proxy for economic uncertainty because of its intuitive appeal and because this measure is available over the longest time period. However, we also perform an extensive sensitivity analysis by considering other commonly used indicators of macroeconomic uncertainty, including the option-implied volatility (Gilchrist et al., 2014b); the cross-sectional dispersion of survey-based forecasts of economic outcome indicators (Bachmann et al., 2013); an economic policy uncertainty index (Baker et al., 2013); and an index of how uncertain agents are about current economic conditions based on economic data releases (Scotti, 2013).

Our results indicate that financial shocks have a significant effect on the economy: A one standard deviation shock to the EBP leads to a modest but persistent increase in uncertainty and an economically and statistically significant decline in economic activity. In addition, the stock market falls substantially, and share prices remain depressed for a considerable period of time. These dynamics are consistent with our maintained assumption that shocks to the EBP are capturing disruptions in the credit-intermediation process and their attendant economic consequences. In the baseline identification scheme, financial shocks account for about one-third of the variation in industrial output, payroll employment, and broad equity valuations at business cycle frequencies, proportions indicating that such shocks are an important source of business cycle fluctuations.

Macroeconomic implications of financial disruptions identified vis-à-vis shocks to the EBP using the alternative identification scheme are very similar: An adverse financial shock induces a broadbased deterioration in economic activity and labor market conditions and leads to a sharp and permanent drop in stock prices. Interestingly, this shock has no effect of the level of uncertainty in the economy under the alternative identification scheme. It is also worth noting that this absence of transmission of financial shocks through changes in economic uncertainty is a data-driven outcome of the analysis and does not reflect a zero restriction as in, for example, a Cholesky identification scheme.

The results also indicate that uncertainty shocks have a meaningful macroeconomic impact in situations where they elicit a significant and persistent response of the EBP—that is, a tightening of financial conditions—as in under the alternative identification, a result consistent with the recent work of Arellano et al. (2012), Christiano et al. (2014), and Gilchrist et al. (2014b). In the baseline identification scheme, by contrast, the impact of uncertainty shocks on economic activity is significantly attenuated and has no effect on financial conditions, even though the identification scheme allows the EBP to respond contemporaneously to an uncertainty innovation. The fact that uncertainty increases upon impact in response to a financial shock also raises the possibility that swings in uncertainty are influenced importantly by changes in financial conditions, suggesting that heightened economic uncertainty may be a symptom, rather than a cause, of financial instability. This empirical result is consistent with the theoretical work of Van Nieuwerburgh and Veldkamp (2006), Fostel and Geanakoplos (2012), and Bachmann and Moscarini (2012), who show that spikes in uncertainty may be the result of a deterioration in macroeconomic conditions rather than an independent source of cyclical fluctuations.

2 Related Literature

By and large, the empirical literature seeking to quantify the macroeconomic impact of financial and uncertainty shocks has treated these two types of disturbances separately. Within the context of VAR models, financial shocks are typically identified using either sign or zero restrictions. For example, Gilchrist and Zakrajšek (2012) assume that shocks to the EBP affect real economic activity and inflation with a lag, but they have a contemporaneous effect on the short- and longterm risk-free rates and stock prices. Using a time-varying parameter VAR model, Prieto et al. (2013) make a similar identifying assumption by allowing shocks to credit spreads to affect real and nominal variables with a lag, while financial variables can react contemporaneously to such a disturbance.

The identification of financial shocks in data-rich environments generally relies on similar timing restrictions. Using security-level data, Gilchrist et al. (2009) construct an array of credit spread portfolios sorted by the issuer's ex-ante expected likelihood of default and the bond's remaining term-to-maturity. By building bond portfolios from the "ground up," they can also construct portfolios of stock returns—sorted by the same default risk categories—corresponding to the firms that issued those bonds. These matched portfolios of stock returns serve as controls for news about firms' future earnings as these borrowers experience shocks to their creditworthiness. Within a structural factor-augmented vector autoregression (FAVAR), Gilchrist et al. (2009) identify financial shocks from the corporate bond spreads that are orthogonal to general measures of economic activity, inflation, real interest rates, and various financial indicators, as well as to equity returns of firms whose outstanding bonds were used to construct credit spreads in the bond portfolios.

In contrast, Boivin et al. (2013) rely on a single credit spread index to identify financial shocks within several large-scale FAVAR specifications. Their identifying assumptions restrict a small subset of endogenous variables from responding contemporaneously to innovations in the representative credit spread. Regardless of the approach used to identify financial shocks, the above papers uniformly find that an unanticipated deterioration in credit conditions—identified vis-àvis a widening of corporate bond credit spreads—leads to substantial and long-lasting declines in economic activity. Moreover, the decomposition of the forecast error variance implies that these financial disruptions account for a sizable fraction of the variation in economic activity at business cycle frequencies.

A related strand of this literature relies on sign restrictions to identify financial shocks. Although the technical details differ from paper to paper, the general idea underlying this approach is that adverse credit supply shocks should lead to a decrease in the quantity of credit, while simultaneously increasing credit spreads or lending rates (see Helbling et al., 2011; Eickmeier and Ng, 2011; Hristov et al., 2012; Gambetti and Musso, 2012; Meeks, 2012; Peersman and Wagner, 2014). Consistent with the literature that relies on timing restrictions to identify credit supply shocks, the overall conclusion from sign-based SVAR identification schemes is the same: The identified exogenous financial shocks are a substantial source of macroeconomic fluctuations, with adverse credit supply shocks inducing a persistent decline in both output and inflation. However, as shown theoretically by Arellano et al. (2012), Christiano et al. (2014), and Gilchrist et al. (2014b), the same sign restrictions will also characterize the response of these variables to an uncertainty shock when financial markets are subject to agency problems.

The empirical literature on uncertainty shocks, in contrast, relies almost exclusively on zero restrictions to trace out the effects of such shocks on the macroeconomy. In his seminal paper, Bloom (2009) identifies uncertainty shocks by placing the VIX index after the stock market return but before the real and nominal blocks of the monthly VAR system. Thus, the transmission of uncertainty shocks through the equity market is ruled out a priori upon impact, and uncertainty innovations have a negligible contemporaneous effect on economic activity. However, once the financial channel opens up—that is, a month after the shock—positive innovations in the VIX index generate a sharp drop in industrial output followed by some overshooting, a pattern consistent with the "wait-and-see" effect of positive uncertainty shocks.³ Bloom et al. (2012) confirm the robustness of this result by considering alternative measures of uncertainty computed from plant-level data—namely, the cross-sectional dispersion of total factor productivity (TFP) and output growth.

In an influential recent paper, Baker et al. (2013) introduce a different notion of uncertainty macroeconomic policy uncertainty. In their monthly VAR specification, the policy uncertainty index is ordered before the stock market return—as well as the real and nominal blocks of the system—thus allowing uncertainty shocks to have a contemporaneous effect on the stock market. According to their results, unanticipated increases in policy uncertainty induce a persistent and significant decline in economic activity, corroborating the earlier results that uncertainty shocks are an important source of cyclical fluctuations.⁴

³Similar results are obtained in the international context by Carrière-Swallow and Cespedes (2013).

⁴In a closely related work, Mumtaz and Surico (2013) consider different sources of macroeconomic policy uncertainty—uncertainty about government spending, taxes, public debt sustainability, and monetary policy—and find that uncertainty about debt sustainability is the most important aspect of policy uncertainty.

Bachmann et al. (2013), in contrast, construct a survey-based measure of uncertainty from the disagreement in business expectations regarding outcome variables such as sales, employment, and hours worked.⁵ Compared with the results of Bloom (2009), their findings indicate that the unanticipated increases in business uncertainty have negligible near-term economic effects, though adverse uncertainty shocks do lead to a persistent and prolonged decline in economic activity. However, once they include a survey-based measure of business confidence in their VAR specifications, the long-run impact of uncertainty innovations ceases to be economically important, and they argue that increases in uncertainty are a reflection of economic downturns rather than an independent driver of cyclical fluctuations.⁶

The evidence from survey-based measures of uncertainty is far more equivocal, however. Leduc and Liu (2014) exploit the fact that some consumer and business surveys tally responses that make explicit references to "uncertainty," insofar it affects the decisions of consumers to purchase big ticket items or firms' decision to expand their capital outlays. Using such direct measures of perceived uncertainty in a small-scale VAR, which, other than a short-term nominal risk-free rate, does not include any other financial variable, they find that the effects of uncertainty shocks on unemployment and inflation are similar to those of standard aggregate demand shocks. Jurado et al. (2013) propose yet another measure of economic uncertainty: A common factor extracted from a panel containing the unforecastable component of a large number of monthly economic and financial indicators. Interestingly, this new measure is only mildly correlated with other commonly used uncertainty proxies. However, when ordered last in a VAR system that includes the aggregate return on equity, the identified uncertainty shocks have large and persistent effects on economic activity.

To the best of our knowledge, there are only two studies that jointly analyze financial and uncertainty shocks in the VAR context. Focusing on the German economy, Popescu and Smets (2010) identify financial and uncertainty shocks using a recursive ordering, in which the uncertainty proxy is placed after the macro block but before the financial market risk index. In other words, they allow uncertainty shocks to elicit an immediate change in financial conditions but not vice versa. Under their identifying assumptions, uncertainty shocks have a small and temporary effects on output and financial risk premia, whereas financial disturbances have a considerably more persistent effect on economic activity.

In light of such restrictive assumptions, Gilchrist et al. (2014b) explore the macroeconomic implications of uncertainty and financial shocks using alternative orderings for the uncertainty and financial stress proxies.⁷ Their first identification scheme posits that uncertainty shocks have an

⁵Although the use of dispersion indexes of expectations—an indication of the degree of disagreement among the survey participants—as a measure of uncertainty has a long tradition in the literature on inflation expectations and inflation uncertainty, this interpretation is not without controversy. In particular, using the individual probabilistic responses from the Survey of Professional Forecasters, D'Amico and Orphanides (2008) show that the disagreement about the mean forecast can be a poor proxy for the actual forecast uncertainty.

⁶Using a weighted-average of realized volatilities across different financial assets and countries as a measure of global uncertainty, Cesa-Bianchi et al. (2014) reach a similar conclusion in a Global VAR (GVAR) framework.

⁷The measure of uncertainty used by Gilchrist et al. (2014b) is based on the volatility of equity returns that have been purged of forecastable variation using a standard empirical asset pricing model. As a result, their uncertainty

immediate impact on the short-term policy rate and corporate bond credit spreads—an indicator of financial market strains—but they affect economic activity and inflation with a lag. In the second scheme, they reverse the ordering of the uncertainty proxy and credit spreads, which allows them to gauge the role of uncertainty shocks conditional on the current state of conditions in credit markets.

The key finding that emerges from their analysis is that the economic significance of uncertainty shocks hinges crucially on whether they have been orthogonalized with respect to the contemporaneous information in credit spreads. This knife-edge result, however, does not apply to financial disturbances identified vis-à-vis unanticipated increases in credit spreads: Under both sets of identifying assumptions, innovations in credit spreads induce a persistent and prolonged decline in the cyclically sensitive components of aggregate output; moreover, in the second scheme, they also elicit a significant increase in uncertainty upon impact.

The implications of these results are twofold. First, they point to distortions in financial markets as an important conduit through which uncertainty shocks can affect macroeconomic outcomes. The fact that uncertainty jumps immediately in response to an adverse credit shock suggest that fluctuations in uncertainty may arise endogenously in response to changes in broad financial market conditions. As noted above, an important goal of this paper is to relax the restrictive timing restrictions implicit in the identification schemes based on recursive ordering in order to analyze jointly the interaction behind changes in financial conditions and economic uncertainty and their effects on the macroeconomy. Using this new framework, we also provide a systematic comparison of the impact of uncertainty shocks across different uncertainty proxies, a useful and timely exercise given the plethora of uncertainty measures available to researchers.

The remainder of the paper is organized along the following lines. Section 3 presents an overview of the uncertainty proxies used in the analysis; it also examines the predictive content of the various proxies for monthly indicators of economic activity, conditional on the excess bond premium. Section 4 introduces the econometric methodology that allows us to explore the interplay of uncertainty and financial shocks without relying on the standard timing or sign restrictions. Section 5 presents the main results from our benchmark specification. In Section 6, we conduct an extensive sensitivity analysis of our main results. Section 7 concludes.

3 Uncertainty, Financial Conditions, and Economic Activity

3.1 Measuring Uncertainty and Financial Conditions

In spite of the intense interest in the role of uncertainty shocks as a source of macroeconomic fluctuations, there is little consensus among economists on what is the best measure of economic uncertainty. As a result, the empirical literature is awash with different uncertainty proxies and new measures crop up all the time. Rather than taking a stand on any particular indicator, we consider six different measures of uncertainty, which we judge span the range of methodological

proxy is less likely to reflect the countercyclical nature of informational and contractual frictions that have been theoretically linked to the countercyclical dispersion of stock returns, a popular measure of economic uncertainty.

	Summary Statistic					
Variable	\mathbf{CV}	α_1	α_2	$lpha_3$	$q_{\scriptscriptstyle LL}$	RVN
RVOL ^a	0.60	0.70	0.14	0.03	-11.14	3.50^{***}
$IVOL^{b}$	0.38	0.64	0.18	0.05	-18.85^{*}	0.08
VXO ^c	0.40	0.83	-0.00	0.04	-16.74	2.25^{**}
$\mathrm{BBD^d}$	0.31	0.84	0.06	0.12	-13.65	0.66
$\mathrm{BES}^{\mathrm{e}}$	0.14	0.80	0.26	0.05	-18.22^{*}	-0.12
$\mathrm{SCOT}^{\mathrm{f}}$	0.39	0.53	0.03	0.07	-15.03	1.18
	Pairwise Correlations					
Variable	RVOL	IVOL	VXO	BBD	BES	SCOT
RVOL	1.00	0.64^{***}	0.84***	0.42^{***}	0.13***	0.37***
IVOL		1.00	0.58^{***}	0.26^{***}	0.23^{***}	0.26^{***}
VXO			1.00	0.41^{***}	0.19^{***}	0.31^{***}
BBD				1.00	-0.01	0.24^{***}
BES					1.00	0.14^{**}
SCOT						1.00

Table 1: Selected Characteristics of Different Uncertainty Measures

NOTE: The entries in the table denote the specified summary statistic: $CV = coefficient of variation; \alpha_k = partial autocorrelation at lag k; q_{LL} = the Elliott and Müller (2006) test statistic of the null hypothesis that the autoregressive coefficients from an AR(3) model are constant over time; and RVN = the Bartels (1982) test statistic of the null hypothesis that the OLS residuals from an AR(3) model are randomly distributed. * <math>p < .05$, and *** p < .01.

^a Realized equity volatility (1973:M1–2012:M12, T = 480).

^b Idiosyncratic equity volatility based on Gilchrist et al. (2014b) (1973:M1–2012:M12, T = 480).

^c Option-implied volatility on the S&P 100 stock futures index (1986:M1–2012:M12, T = 324).

^d Uncertainty measure based on Baker et al. (2013) (1985:M1–2012:M12, T = 336).

^e Uncertainty measure based on Bachmann et al. (2013) (1973:M1–2011:M12, T = 468).

^f Uncertainty measure based on Scotti (2013) (1991:M1–2012:M12, T = 264).

approaches used to infer fluctuations in economic uncertainty.

We focus on measures available at a monthly frequency. Our benchmark proxy for economic uncertainty is the realized stock market volatility (**RVOL**), a measure that is simple to construct and is available over the longest time period. We also consider two other measures based on equity valuations: shocks to the idiosyncratic volatility of (excess) equity returns that are common to U.S. nonfinancial firms (**IVOL**), a measure proposed by Gilchrist et al. (2014b); and the option-implied volatility on the S&P 100 stock futures index constructed by the Chicago Board of Option Exchange (**VXO**).⁸

The remaining three proxies eschew financial market data. Specifically, our fourth proxy is the widely cited index of policy-related economic uncertainty developed by Baker et al. (2013) (**BBD**). The fifth proxy for macroeconomic uncertainty, put forth by Bachmann et al. (2013), is a measure

⁸We use the VXO, as opposed to the VIX option-implied volatility, because the VXO is available starting in January 1986, compared with January 1990, the starting date for the VIX. The correlation between these two indicators, however, is almost 0.99 at a monthly frequency.

of forecast dispersion constructed using the Philadelphia Fed's Business Outlook Survey (**BES**). And lastly, we also consider the real-time uncertainty index developed by Scotti (2013), who, using a dynamic factor model and "surprises" derived from economic data releases, infers how uncertain are financial market participants about the economic outlook (**SCOT**). (Details concerning the construction of the various uncertainty measures are contained in Appendix A.)

The top panel of Table 1 provides some summary statistics for the six uncertainty measures used in our analysis. Not too surprising, the coefficients of variation for the uncertainty proxies derived from equity valuations tend to be larger compared with those of uncertainty proxies that are not based on financial market data. As evidenced by the partial correlations, most of the measures exhibit significant positive first-order autocorrelation; the two exceptions are the IVOL and SCOT uncertainty proxies, which are governed by processes with much lower serial correlation. The degree of serial dependence, however, dies off very quickly in every case. Letting each series follow an AR(3) process, there appears to be little evidence of parameter instability in the autoregressive coefficients, according to the Elliott and Müller (2006) Quasi-Local Level test: Only for the IVOL and BES uncertainty measures, we reject—and only at the 10 percent significance level—the null hypothesis that the autoregressive coefficients are constant over time. Moreover, in most cases, the resulting residuals appear to be distributed randomly.

As shown by the entries in the bottom panel, the six series, in general, exhibit significant positive contemporaneous correlation. As expected, the highest degree of comovement is between the three uncertainty proxies based on the stock market data (RVOL, IVOL, and VXO). The other three measures (BBD, BES, and SCOT) are also positively correlated with their equity-based counterparts, though to a significantly lesser extent. The pairwise correlations between BBD, BES, and SCOT, in contrast, are notably smaller. In fact, the correlation between the BBD and BES proxies is statistically and economically indistinguishable from zero, while the correlation between the BES and SCOT measures, though positive, is quite low.

3.2 Measuring Financial Conditions

As discussed above, we rely on the information contained in corporate credit spreads to measure strains in financial markets. In particular, we use the excess bond premium, an estimate of the extra compensation demanded by bond investors for bearing exposure to U.S. nonfinancial corporate credit risk, above and beyond the compensation for expected losses. As emphasized by Gilchrist and Zakrajšek (2012), the U.S. corporate cash market is served by major financial institutions and fluctuations in the EBP thus capture shifts in the risk attitudes of these institutions and their willingness to bear credit risk and to intermediate credit more generally in global financial markets.⁹

The solid line in Figure 2 shows this indicator of the effective "risk-bearing capacity" of the

⁹This interpretation is also supported by the empirical work of Adrian and Shin (2010) and Adrian et al. (2010a,b), who show that risk premiums in asset markets are very sensitive to movements in capital and balance sheet conditions of financial intermediaries. Theoretical foundations for such "intermediary" asset pricing theories are developed in the influential work of He and Krishnamurthy (2013) and Brunnermeier and Sannikov (2014).



Figure 2: Financial Conditions and Uncertainty

NOTE: Sample period: monthly data from 1973:M1 to 2012:M12. The solid line depicts the estimate of the excess bond premium (EBP), an indicator of the tightness of financial conditions (see Gilchrist and Zakrajšek, 2012). The dotted line shows the realized stock market volatility (RVOL), our benchmark measure of economic uncertainty. The shaded vertical bars denote the NBER-dated recessions.

financial intermediary sector over the 1973–2012 period. Note that the EBP appears to be a particularly timely indicator of strains in the financial system. As the intensifying downturn in the U.S. subprime mortgage market in the latter half of 2006 led to the emergence of significant strains in term funding markets in the United States and Europe during the summer of 2007, the EBP rose sharply, an increase concomitant with a jump in the RVOL measure of uncertainty (dotted line).

The extent of covariation between financial conditions and economic uncertainty is detailed more formally in Table 2, which shows the pairwise cross-correlations between the EBP and the different uncertainty proxies at various leads and lags. The cross-correlations are all positive and are largest contemporaneously (h = 0), ranging from a low of about 0.1 (BES) to a high of almost 0.6 (VXO). Moreover, there is a substantial degree of comovement between the EBP and various proxies at different near-term horizons, underscoring the close relationship between changes in financial conditions and swings in economic uncertainty over the course of a business cycle.

Note that this holds true even for uncertainty proxies that are not explicitly derived from financial market data, such as the BBD and SCOTT measures. However, one clear exception to this pattern is the BES uncertainty proxy—a measure based on the survey respondents' disagreement regarding future economic outcomes—the past values of which appear to be largely uncorrelated with future fluctuations in the EBP; on the other hand, a tightening of financial conditions is mildly

Lag/Lead (h)	RVOL	IVOL	VXO	BBD	BES	SCOT
-3	0.44^{***}	0.16^{***}	0.53^{***}	0.36^{***}	0.01	0.38^{***}
-2	0.48^{***}	0.19^{***}	0.55^{***}	0.36^{***}	0.06	0.39^{***}
-1	0.54^{***}	0.24^{***}	0.58^{***}	0.35^{***}	0.07	0.39^{***}
0	0.53^{***}	0.24^{***}	0.59^{***}	0.35^{***}	0.09^{**}	0.37^{***}
1	0.54^{***}	0.25^{***}	0.59^{***}	0.32^{***}	0.09^{**}	0.37^{***}
2	0.48^{***}	0.23^{***}	0.52^{***}	0.26^{***}	0.10^{**}	0.35^{***}
3	0.46^{***}	0.22^{***}	0.48^{***}	0.24^{***}	0.10^{**}	0.35^{***}

Table 2: Cross-Correlations Between the EBP and Various Uncertainty Measures

NOTE: The entries in the table denote the cross-correlations between the EBP in month t and the specified uncertainty measure at month t + h. RVOL = realized equity volatility (1973:M1–2012:M12, T = 480); IVOL = idiosyncratic equity volatility based on Gilchrist et al. (2014b) (1973:M1–2012:M12, T = 480); VXO = option-implied volatility on the S&P 100 stock futures index (1986:M1–2012:M12, T = 324); BBD = uncertainty measure based on Baker et al. (2013) (1985:M1–2012:M12, T = 336); BES = uncertainty measure based on Bachmann et al. (2013) (1973:M1–2011:M12, T = 468); and SCOT = uncertainty measure based on Scotti (2013) (1991:M1–2012:M12, T = 264). * p < .05, and *** p < .01.

indicative of a future increase in this uncertainty indicator. All told, these results highlight the need of an identification scheme that allows financial conditions to respond contemporaneously to an uncertainty shock, while at the same time allowing economic uncertainty to respond immediately to a change in credit supply conditions prompted by a financial shock.

3.3 Uncertainty and Financial Conditions as Predictors of Economic Activity

To explore more formally the relative roles of uncertainty and financial conditions as predictors of the near-term course in economic activity, we first perform a simple forecasting exercise. Letting Y_t denote a monthly indicator of economic activity, we estimate the following univariate specification:

$$\Delta_h Y_{t+h} = \alpha + \beta_1 \sigma_t + \beta_2 \text{EBP}_t + \gamma' \mathbf{Z}_t + \sum_{i=1}^p \rho_i \Delta Y_{t-i} + \epsilon_{t+h}, \qquad (1)$$

where $\Delta_h Y_{t+h} \equiv \frac{1200}{h+1} \ln \left(\frac{Y_{t+h}}{Y_{t-1}} \right)$ and $h \ge 0$ is the forecast horizon ($\Delta_0 \equiv \Delta$). In the forecasting regression (1), σ_t denotes an uncertainty indicator in month t, EBP_t is the excess bond premium in the same month, \mathbf{Z}_t is a vector of other financial indicators that may contain useful information about the economic outlook, and ϵ_{t+h} is the forecast error.¹⁰

Our baseline forecasting exercise abstracts from the information content of other financial indicators—that is, $\mathbf{Z}_t = \emptyset$. For each of the six uncertainty indicators, we estimate the resulting specification (1) separately by OLS. Measures of monthly economic activity considered include (manufacturing) industrial production and private (nonfarm) payroll employment. To facilitate

¹⁰Note that the timing implied by this specification allows for the possibility of "nowcasting" (i.e., h = 0), and it is intended to capture the fact that when forecasting an indicator of economic activity in period t, economists, because of reporting lags, typically do not observe the current value of the indicator, while the financial indicators used to assess the current state of credit market conditions and most uncertainty indicators are readily available.

the comparison of the predictive content of different uncertainty measures and the EBP, we report the standardized estimates of the coefficients β_1 and β_2 . The results for the forecast horizon of 3 months are tabulated in panels A and B of Table 3, whereas those for the 12-month-ahead horizon are shown in panels C and D.¹¹

According to Table 3, the predictive content of various uncertainty measures for economic activity is fairly uneven at the 3-month-ahead forecast horizon (panels A and B). While increases in the equity-based measures of economic uncertainty are generally associated with statistically and economically significant slowdown in the growth of industrial output and employment, the information content of the VXO is quite limited. Similarly, both the BBD policy uncertainty index and the SCOT index appear to be uninformative about the near-term course of economic activity. The survey-based BES uncertainty index, by contrast, is a robust predictor of the growth in both industrial output and employment—an increase of one standard deviation in the BES uncertainty index in month t is estimated to shave off about two percentage points in the (annualized) growth rate of industrial production over the subsequent three months and nearly three-quarters of a percentage point in the growth of nonfarm payroll employment.

In comparison with these uncertainty measures, the EBP appears to be a far more robust predictor of near-term economic developments. Moreover, the magnitude of the estimated coefficients implies an economically significant negative relationship between changes in financial conditions and future economic activity. For example, using a coefficient estimate of 0.35, a value in the range of the central tendency of the point estimates reported in panel A, an increase of 50 basis points in the EBP in month t (about one standard deviation) implies a 2.5 percentage points (annualized) drop in the growth rate of industrial output over the subsequent three months.

The uneven forecasting performance of the uncertainty indicators is also evident at longer horizons (panels C and D). While increases in the RVOL, IVOL, and BES indicators continue to signal a slowdown in economic activity, a rise in the BBD policy uncertainty index is associated with an economically and statistically significant acceleration in industrial production and a pickup in employment growth over the subsequent year. The same is true of the SCOT uncertainty index, though the magnitude of the estimated effect is economically negligible in that case. Movements in the EBP, in contrast, continue to provide unambiguous and informative signals about the evolution of the year-ahead economic outlook, with a tightening of financial conditions portending a marked deceleration in industrial output and a significant deterioration in labor market conditions.

Our next forecasting exercise examines the robustness of the above results by conditioning on other potentially useful information in financial markets. Specifically, in this alternative specification we let $\mathbf{Z}_t = [TS_t \ RR_t]'$, where TS_t denotes the "term spread" and RR_t is the real 2-year

¹¹To take into account the MA(h+1) structure of the error term ϵ_{t+h} induced by overlapping observations, we compute the covariance matrix of the estimated parameters according to Hodrick (1992). As shown by Ang and Bekaert (2007), the standard errors developed by Hodrick (1992) retain the correct size even in relatively small samples when testing the null of no predictability in the context of overlapping observations. In all specifications, we set the lag length p in equation (1) equal to 6. All the results, however, are completely robust to both shorter (p = 3) and longer (p = 12) lag lengths.

Predictor	RVOL	IVOL	VXO	BBD	BES	SCOT	
A. Industrial Production $(h = 3 \text{ months})$							
σ_t	-0.179	-0.111	-0.018	-0.031	-0.270	-0.068	
	[4.31]	[3.23]	[0.34]	[0.78]	[6.49]	[1.30]	
EBP_t	-0.278	-0.339	-0.401	-0.397	-0.368	-0.498	
	[6.23]	[7.11]	[6.42]	[6.75]	[7.59]	[7.54]	
Adj. R^2	0.352	0.341	0.467	0.457	0.398	0.536	
B. Payroll Employment $(h = 3 \text{ months})$							
σ_t	-0.157	-0.130	-0.054	0.006	-0.210	-0.049	
C C	[7.15]	[4.74]	[1.65]	[0.25]	[4.76]	[2.05]	
EBP_t	-0.165	-0.216	-0.213	-0.235	-0.222	-0.337	
Ū	[6.60]	[8.21]	[6.33]	[6.96]	[8.22]	[8.51]	
Adj. R^2	0.663	0.662	0.806	0.797	0.684	0.839	
C. Industrial Production $(h = 12 \text{ months})$							
σ_t	-0.119	-0.055	0.041	0.121	-0.357	0.040	
	[4.58]	[1.82]	[1.13]	[4.04]	[10.08]	[2.12]	
EBP_t	-0.295	-0.340	-0.473	-0.483	-0.351	-0.550	
	[7.81]	[8.98]	[9.82]	[10.17]	[9.40]	[10.44]	
Adi B^2	0 160	0.153	0.241	0.242	0 266	0 290	
	1 (1	10	0.211	0.212	0.200	0.200	
D. Payroll Employment $(h = 12 \text{ months})$							
σ_t	-0.164	-0.175	0.011	0.147	-0.349	0.037	
	[10.68]	[7.77]	[0.36]	[8.26]	[8.98]	[3.32]	
EBP_t	-0.251	-0.298	-0.330	-0.359	-0.299	-0.463	
	[12.19]	[14.41]	[11.69]	[12.52]	[14.56]	[15.34]	
Adj. R^2	0.420	0.429	0.602	0.606	0.503	0.644	

Table 3: Uncertainty, Financial Conditions, and Economic Activity (Baseline Specification)

NOTE: The dependent variable in each specification is $\Delta_h Y_{t+h}$, the annualized growth rate in the specified indicator of economic activity from month t - 1 to month t + h. The entries in the row of the table corresponding to σ_t denote the standardized estimates of the OLS coefficients associated with the specified uncertainty indicator in month t: RVOL = realized equity volatility (1973:M1-2012:M12, T = 480); IVOL = idiosyncratic equity volatility based on Gilchrist et al. (2014b) (1973:M1-2012:M12, T = 480); VXO = option-implied volatility on the S&P 100 stock futures index (1986:M1-2012:M12, T = 324); BBD = uncertainty measure based on Baker et al. (2013) (1985:M1-2012:M12, T = 336); BES = uncertainty measure based on Bachmann et al. (2013) (1973:M1-2011:M12, T = 468); and SCOT = uncertainty measure based on Scotti (2013) (1991:M1-2012:M12, T = 264). The entries in the row of the table corresponding to EBP_t denote the standardized estimates of the OLS coefficients associated with the excess bond premium in month t, an indicator of the tightness of financial conditions (see Gilchrist and Zakrajšek, 2012). In addition to σ_t and EBP_t, each specification also includes a constant and 6 lags of ΔY_{t-1} (not reported). Absolute asymptotic t-statistics reported in brackets are computed according to Hodrick (1992).

Treasury yield.¹² The results of this exercise are reported in Table 4. To conserve space, we report the results for h = 12 only, but the results for the shorter-term forecast horizons were qualitatively very similar.

As shown in panel A, the inclusion of the slope of the yield curve and the real short-term rate—two indicators of the stance of monetary policy—significantly improves the in-sample fit of the regression for the year-ahead growth in industrial production. For the specifications involving RVOL and IVOL, which are estimated over the longest sample period (1973:M1–2012:M12), the adjusted R^2 more than doubles; for the specifications estimated using data since the mid-1980s—those involving the BBD and VXO measures of uncertainty—the increase in the goodness-of-fit is not as large, though it is still substantial. While the term spread and the real short-term rate both contain significant predictive power for the year-ahead growth in industrial output, conditioning on these two financial indicators in no way affects the information content of the EBP. In contrast, the forecasting ability of the different uncertainty indicators, which was already uneven according to the baseline specification, diminishes notably for the most informative indicators (RVOL and BES).

A similar picture emerges when we look at the employment growth (panel B). The inclusion of the term spread and the real short-term rate in the predictor set yields sizable improvements in the in-sample fit, though not as pronounced as in the case of industrial production, a much more volatile measure of economic activity. The economic significance of the most informative uncertainty indicators—RVOL, IVOL, and BES, according to the baseline specification—tends to diminish somewhat with the inclusion of these two financial indicators, while the economic impact of the EBP is unaffected across all specification. Note that even conditional on this richer information set, increases in the BBD policy uncertainty index continue to signal a pickup in economic activity. Increases in the EBP, by contrast, continue to be unambiguously associated with a significant deterioration in the medium-term economic outlook.

The results from the above forecasting exercises—which, of course, are completely silent on the causal relationship between uncertainty, financial conditions, and economic activity—are instructive for two reasons. First, they underscore the fact that measures of financial distress based on corporate bond credit spreads are highly informative about the economic outlook. Second, although the information content of various uncertainty indicators appears to be somewhat mixed, the available evidence nevertheless suggests that some of these indicators contain economically and statistically significant marginal predictive power for economic activity. In combination with the fact that episodes of financial distress are closely associated with periods of heightened economic uncertainty and that theoretical mechanisms based on frictions in financial markets imply an important interaction between changes in financial conditions and fluctuations in uncertainty, a natural question raised by this evidence concerns the relative importance of financial and uncer-

¹²The term spread is defined as the difference between the 3-month constant-maturity Treasury yield and the 10-year constant-maturity yield. The real 2-year Treasury yield is defined as the 2-year nominal Treasury yield in month t less realized inflation, where realized inflation is given by the log-difference between the core consumer price index in period t-1 and its lagged value a year earlier.

Predictor	RVOL	IVOL	VXO	BBD	BES	SCOT		
A. Industrial Production $(h = 12 \text{ months})$								
σ_t	-0.048	-0.063	0.055	0.162	-0.241	-0.007		
	[1.97]	[2.13]	[1.50]	[4.24]	[6.37]	[0.38]		
EBP_t	-0.393	-0.402	-0.537	-0.559	-0.411	-0.541		
	[10.49]	[10.72]	[10.99]	[11.07]	[11.19]	[10.70]		
TS_t	0.417	0.421	0.357	0.299	0.342	0.361		
	[11.29]	[11.66]	[9.86]	[8.24]	[8.51]	[8.95]		
RR_t	0.222	0.226	0.400	0.432	0.240	0.413		
	[5.15]	[5.31]	[8.36]	[8.49]	[5.74]	[8.07]		
0								
Adj. R^2	0.387	0.389	0.417	0.420	0.433	0.461		
B. Payroll Empl	oyment ($h =$	= 12 months)					
σ_t	-0.125	-0.183	0.009	0.144	-0.269	0.000		
	[7.35]	[8.17]	[0.36]	[6.66]	[6.94]	[0.04]		
EBP_t	-0.277	-0.309	-0.347	-0.391	-0.316	-0.426		
	[13.14]	[14.98]	[12.44]	[13.66]	[15.79]	[13.99]		
TS_t	0.321	0.330	0.342	0.285	0.253	0.323		
	[12.63]	[13.03]	[12.45]	[10.44]	[9.53]	[10.93]		
RR_t	0.092	0.108	0.251	0.299	0.114	0.235		
	[2.26]	[2.76]	[10.42]	[11.68]	[2.99]	[8.97]		
Adi B^2	0.529	0.550	0 711	0 716	0.576	0.732		

Table 4: Uncertainty, Financial Conditions, and Economic Activity (Augmented Specification)

NOTE: The dependent variable in each specification is $\Delta_{12}Y_{t+12}$, the annualized growth rate in the specified indicator of economic activity from month t-1 to month t+12. The entries in the row of the table corresponding to σ_t denote the standardized estimates of the OLS coefficients associated with the specified uncertainty indicator in month t: RVOL = realized equity volatility (1973:M1-2012:M12, T = 480); IVOL = idiosyncratic equity volatility based on Gilchrist et al. (2014b) (1973:M1-2012:M12, T = 480); VXO = option-implied volatility on the S&P 100 stock futures index (1986:M1-2012:M12, T = 324); BBD = policy uncertainty measure based on Baker et al. (2013) (1985:M1-2012:M12, T = 336); BES = uncertainty measure based on Bachmann et al. (2013) (1973:M1-2011:M12, T = 468); and SCOT = uncertainty measure based on Scotti (2013) (1991:M1-2012:M12, T = 264). The entries in the row of the table corresponding to EBP_t denote the standardized estimates of the OLS coefficients associated with the excess bond premium in month t, an indicator of the tightness of financial conditions (see Gilchrist and Zakrajšek, 2012). The entries in the rows of the table corresponding to TS_t and RR_t denote the standardized estimates of the OLS coefficients associated with the 3m/10y term spread and the real 2-year Treasury yield, respectively. In addition to σ_t , EBP_t, TS_t , and RR_t , each specification also includes a constant and 6 lags of ΔY_{t-1} (not reported). Absolute asymptotic t-statistics reported in brackets are computed according to Hodrick (1992).

tainty shocks in business cycle fluctuations. To answer this question empirically, however, one has to take a stand on the joint identification of these two types of shocks.

4 Identifying Uncertainty and Financial Shocks

To identify uncertainty and financial shocks, we employ the penalty function approach (PFA) proposed initially by Faust (1998) and Uhlig (2005) in the VAR-based identification of monetary policy shocks, an approach that was later extended by Mountford and Uhlig (2009) to jointly identify multiple structural disturbances. The PFA selects a structural VAR model by maximizing a criterion function subject to inequality constraints. The criterion function consists of the sum of impulse response functions (IRFs) of selected variables at horizons 0 to H, while the inequality constraints correspond to sign restrictions on these IRFs. In this section, we first provide a general formulation of the PFA. We then discuss the rationale underlying our two identification schemes—that is, the choice of the penalty functions—and the estimation details.

4.1 The SVAR and the Penalty Function

Consider the following SVAR:

$$\mathbf{y}_{t}'\mathbf{A}_{0} = \sum_{i=1}^{p} \mathbf{y}_{t-i}'\mathbf{A}_{i} + \mathbf{c} + \boldsymbol{\epsilon}_{t}'; \quad t = 1, \dots, T,$$
(2)

where \mathbf{y}_t is an $n \times 1$ vector of endogenous variables, $\boldsymbol{\epsilon}_t$ is an $n \times 1$ vector of structural shocks, \mathbf{A}_i , $i = 1, \ldots, p$, is an $n \times n$ matrix of structural parameters with \mathbf{A}_0 invertible, \mathbf{c} is a $1 \times n$ vector of parameters, p is the lag length, and T is the sample size. Conditional on past information and the initial conditions $\mathbf{y}_0, \ldots, \mathbf{y}_{1-p}$, the vector of structural shocks $\boldsymbol{\epsilon}_t$ is assumed to be Gaussian with mean zero and covariance matrix \mathbf{I}_n , the $n \times n$ identity matrix.

The SVAR model in equation (2) can be written more compactly as

$$\mathbf{y}_t' \mathbf{A}_0 = \mathbf{x}_t' \mathbf{A}_+ + \boldsymbol{\epsilon}_t',\tag{3}$$

where $\mathbf{A}'_{+} = [\mathbf{A}'_{1} \dots \mathbf{A}'_{p} \mathbf{c}']$ and $\mathbf{x}'_{t} = [\mathbf{y}'_{t-1} \dots \mathbf{y}'_{t-p} \mathbf{1}]$. The dimension of \mathbf{A}_{+} is $m \times n$, where m = np + 1, and the elements of matrices \mathbf{A}_{0} and \mathbf{A}_{+} correspond to the structural parameters of the VAR system. The reduced-form representation of the VAR is given by

$$\mathbf{y}_t' = \mathbf{x}_t' \mathbf{B} + \mathbf{u}_t'$$

where $\mathbf{B} = \mathbf{A}_{+}\mathbf{A}_{0}^{-1}$, $\mathbf{u}'_{t} = \boldsymbol{\epsilon}'_{t}\mathbf{A}_{0}^{-1}$, and $\mathbb{E}[\mathbf{u}_{t}\mathbf{u}'_{t}] = (\mathbf{A}_{0}\mathbf{A}'_{0})^{-1} \equiv \boldsymbol{\Sigma}$; the matrices \mathbf{B} and $\boldsymbol{\Sigma}$ are the reduced-form parameters. In this context, the impulse response functions are defined as follows:

Definition 1 Let $\{\mathbf{A}_0, \mathbf{A}_+\}$ represent arbitrary structural parameters. Then, the IRF of the *i*-th variable to the *j*-th structural shock at a finite horizon h—denoted by $\mathbf{L}_h(\mathbf{A}_0, \mathbf{A}_+)_{ij}$ —corresponds

to the element in row i and column j of the matrix $\begin{bmatrix} \mathbf{A}_0^{-1} \mathbf{J}' \mathbf{F}^h \mathbf{J} \end{bmatrix}'$, where

$$\mathbf{F} = \begin{bmatrix} \mathbf{A}_1 \mathbf{A}_0^{-1} & \mathbf{I}_n & \cdots & \mathbf{0} \\ \vdots & \vdots & \ddots & \vdots \\ \mathbf{A}_{p-1} \mathbf{A}_0^{-1} & \mathbf{0} & \cdots & \mathbf{I}_n \\ \mathbf{A}_p \mathbf{A}_0^{-1} & \mathbf{0} & \cdots & \mathbf{0} \end{bmatrix} \quad and \quad \mathbf{J} = \begin{bmatrix} \mathbf{I}_n \\ \mathbf{0} \\ \vdots \\ \mathbf{0} \end{bmatrix}.$$

Note that the matrix of IRFs upon impact is given by $\mathbf{L}_0(\mathbf{A}_0, \mathbf{A}_+) = \mathbf{A}_0^{-1}$. As in Uhlig (2005), we characterize the set of all possible IRFs using an $n \times n$ orthonormal matrix $\mathbf{Q} \in \mathcal{O}(n)$, where $\mathcal{O}(n)$ denotes the set of all orthonormal $n \times n$ matrices. To see this, let \mathbf{T} denote the lower triangular matrix from the Cholesky factorization of Σ . Then for any orthonormal matrix \mathbf{Q} , the matrix $\tilde{\mathbf{A}}_0^{-1} = \mathbf{T}\mathbf{Q}$ is also a decomposition of Σ that satisfies $[\tilde{\mathbf{A}}_0\tilde{\mathbf{A}}_0']^{-1} = \Sigma$. Identification of the SVAR thus amounts to specifying a set of restrictions on the matrix \mathbf{Q} . For instance, $\mathbf{Q} = \mathbf{I}_n$ imposes identification based on the recursive ordering of the VAR (i.e., the widely used Cholesky decomposition), while sign restrictions on the IRFs involve specifying a set of admissible \mathbf{Q} matrices.

It is worth emphasizing that our identification strategy does not identify all n structural shocks—that is, the entire \mathbf{Q} matrix. Rather, we identify a subset $k \leq n$ of shocks, represented by $\mathbf{q}_j = \mathbf{Q}\mathbf{e}_j$, j = 1, ..., k, where \mathbf{e}_j denotes the j-th column of \mathbf{I}_n . Specifically, let $\{\mathbf{A}_0, \mathbf{A}_+\}$ be any draw of the structural parameters and consider a case where the identification of the j-th structural shock restricts the IRF of a set of variables indexed by $I_j^+ \subset \{0, 1, ..., n\}$ to be positive and the IRF of a set of variables indexed by $I_j^- \subset \{0, 1, ..., n\}$ to be negative. Furthermore, assume that the restrictions on variable i are enforced for $H \geq 0$ periods. The identification of \mathbf{q}_j then amounts to solving the following optimization problem:

$$\mathbf{q}_{j}^{*} = \underset{\mathbf{q}_{j}}{\operatorname{arg\,min}} \Psi(\mathbf{q}_{j})$$
(4)
s. t.

$$\mathbf{e}_{i}^{\prime}\mathbf{L}_{h}(\mathbf{T}^{-1},\mathbf{B}\mathbf{T}^{-1})\mathbf{q}_{j} > 0, \quad i \in I_{j}^{+} \text{ and } h = 0,\ldots,H;$$

$$(5)$$

$$\mathbf{e}'_{i}\mathbf{L}_{h}(\mathbf{T}^{-1},\mathbf{B}\mathbf{T}^{-1})\mathbf{q}_{j} < 0, \quad i \in I_{j}^{-} \text{ and } h = 0,\ldots,H;$$

$$(6)$$

$$\mathbf{Q}_{i-1}^{*\prime}\mathbf{q}_{i} = 0, \tag{7}$$

where

$$\Psi(\mathbf{q}_j) = \sum_{i \in I_j^+} \sum_{h=0}^{H} \left(-\frac{\mathbf{e}_i' \mathbf{L}_h(\mathbf{T}^{-1}, \mathbf{B}\mathbf{T}^{-1}) \mathbf{q}_j}{\sigma_i} \right) + \sum_{i \in I_j^-} \sum_{h=0}^{H} \left(\frac{\mathbf{e}_i' \mathbf{L}_h(\mathbf{T}^{-1}, \mathbf{B}\mathbf{T}^{-1}) \mathbf{q}_j}{\sigma_i} \right),$$
(8)

 σ_i is the standard deviation of variable *i*, and $\mathbf{Q}_{j-1}^* = [\mathbf{q}_1^* \dots \mathbf{q}_{j-1}^*]$, for $j = 1, \dots, n$.¹³

In line with the existing literature, our characterization of the penalty function (8) assumes

¹³We follow the convention by letting \mathbf{Q}_0^* equal the $n \times n$ null matrix; to obtain σ_i , we compute the standard deviation of the OLS residuals associated with the *i*-th variable.

that the vector \mathbf{q}_j rotates the IRFs associated with the Cholesky factor matrix **T**. As in Uhlig (2005) and Mountford and Uhlig (2009), the constraints (5) and (6) correspond to sign restrictions on the IRFs that enter the penalty function $\Psi(\mathbf{q}_j)$. Note that the PFA selects a unique rotation matrix \mathbf{Q}^* for each draw of the structural parameters $\{\mathbf{A}_0, \mathbf{A}_+\}$, the same as in the standard point identification. However, the PFA differs from the standard sign restrictions approach because the IRFs corresponding to each draw of the structural parameters $\{\mathbf{A}_0, \mathbf{A}_+\}$ are computed using the rotation matrix \mathbf{Q}^* that minimizes the penalty function (8). The constraints (5) and (6) do not identify a set of structural models—that is, a set of \mathbf{Q} matrices—rather, they define a set of admissible rotation matrices from which the \mathbf{Q}^* is selected. In our implementation of the PFA, however, the sign restrictions are never violated, and the rotation matrix \mathbf{Q}^* always lies in the interior of the admissible set.

Following Mountford and Uhlig (2009), we jointly identify multiple structural shocks sequentially by specifying a penalty function (8) for each shock and then imposing—via the constraint (7) in the optimization problem (4)—that shock j is orthogonal to shocks $1, \ldots, j-1$. This sequential approach is reminiscent of a recursive ordering implicit in the Cholesky decomposition—in fact, this approach returns a Cholesky factorization of Σ if the penalty function that identifies shock jcontains only the impact response (H = 0) of variable j, for $j = 1, \ldots, n$. In all other cases, the sequential identification of the shocks using the PFA does not impose any zero restrictions on the structural parameters { $\mathbf{A}_0, \mathbf{A}_+$ } or on the IRFs at any horizon.

This identification strategy, however, is not invariant to the ordering of the shocks. Next we describe the two identification schemes used in our analysis, which share the same penalty functions and differ only in the ordering of the uncertainty and financial shocks. For the purpose of describing the two identification schemes, it suffices to say that without loss of generality, we order the EBP and an uncertainty measure first and second, respectively, in the vector of the endogenous variables \mathbf{y}_t .

4.2 Identification and Estimation

Our baseline identification scheme involves two restrictions. First, a financial shock corresponds to an innovation that generates the *largest increase* in the EBP for the first six months. The penalty function associated with this shock is given by

$$\Psi(\mathbf{q}_1) = \sum_{h=0}^{5} \left(-\frac{\mathbf{e}_1' \mathbf{L}_h(\mathbf{T}^{-1}, \mathbf{B}\mathbf{T}^{-1})\mathbf{q}_1}{\sigma_1} \right),$$

with

$$\mathbf{e}_{1}'\mathbf{L}_{h}(\mathbf{T}^{-1},\mathbf{B}\mathbf{T}^{-1})\mathbf{q}_{1}>0, \quad h=0,\ldots,5,$$

where j = 1 because we identify the first shock in the system; i = 1 because the EBP is the first variable in the system; $I_1^+ = \{1\}$ and $I_1^- = \emptyset$ because we only impose positive restrictions on the EBP; and H = 5 because we restrict the IRFs of the EBP for six month (in our notation the impact response occurs in period 0).

The second restriction identifies uncertainty shocks. Specifically, an uncertainty shock is an innovation that generates the *largest increase* in a measure of uncertainty for the first six months and is orthogonal to the financial disturbance identified by the first restriction. The penalty function associated with this shock is given by

$$\Psi(\mathbf{q}_2) = \sum_{h=0}^5 \left(-\frac{\mathbf{e}_2' \mathbf{L}_h(\mathbf{T}^{-1}, \mathbf{B}\mathbf{T}^{-1})\mathbf{q}_2}{\sigma_2} \right),$$

with

$$\mathbf{e}_{2}'\mathbf{L}_{h}(\mathbf{T}^{-1}, \mathbf{B}\mathbf{T}^{-1})\mathbf{q}_{2} > 0, \quad h = 0, \dots, 5;$$

 $\mathbf{Q}_{1}^{*'}\mathbf{q}_{2} = 0,$

where j = 2 because we are identifying the second shock and i = 2 because uncertainty is the second variable in the VAR.

Implicit in this identification scheme is an economically plausible argument that an unanticipated worsening in financial conditions is due primarily to an adverse financial shock, while the exogenous fluctuations in economic uncertainty are driven mainly by uncertainty shocks. Importantly, these identifying assumptions do not rule out the possibility that economic uncertainty might react contemporaneously to a change in financial conditions induced by a financial shock; by the same token, financial conditions are allowed to change with the impact of an uncertainty shock.

Although more general than the identification strategy based on the recursive ordering of the VAR system, our baseline identification scheme still imposes a particular form of a timing restriction—namely, the optimization problem that identifies the uncertainty shock is solved conditional on solving the optimization that identifies the financial shock. Given the close relationship between changes in financial conditions and swings in economic uncertainty, this sequential ordering may lead to a concern that the identified financial shocks are to some extent contaminated by uncertainty shocks.

To take into account this possibility, our alternative identification scheme reverses the ordering of the two optimization problems. In other words, we first identify an uncertainty shock as an innovation that generates the largest increase in the uncertainty indicator for the first six months after its impact. In the second step, a financial shock—orthogonal to the uncertainty shock implied by the first step—is identified as shock that induces the largest increase in the EBP for the first six months after its impact. We readily acknowledge that neither scheme fully resolves this thorny identification problem. However, in the absence of valid external instruments, we view the two approaches as providing useful bounds on the role of uncertainty and financial shocks in business cycle fluctuations.

In spirit, our implementation of the PFA is similar to the identification strategy used by Uhlig (2003), Barsky and Sims (2011), and Kurmann and Otrok (2013), all of whom identify structural shocks by maximizing—over a pre-specified forecast horizon—the shock's contribution to the fore-

cast error variance of a given variable. Our identification scheme, in contrast, identifies shocks by maximizing the impulse response of a given variable over a pre-specified horizon. We chose this approach because it implies that the identified financial shocks generate a persistent increase in the EBP—a period of financial distress—while uncertainty shocks lead to a persistent increase in an uncertainty proxy, that is, a period of heightened uncertainty. Selecting shocks that maximize the forecast error variance of these two variables, by contrast, does not guarantee that the their IRFs will not switch signs over the forecast horizon. However, the results reported below indicate that the two shocks identified using the PFA also explain a large fraction of the forecast error variance of their respective variables at business cycle frequencies, which suggests that the two approaches are unlikely to yield very different conclusions.

To implement the two identification schemes, we employ Bayesian estimation techniques. The benchmark monthly VAR specification consists of nine endogenous variables: (1) the EBP; (2) the RVOL uncertainty indicator; (3) the log of manufacturing industrial production; (4) the log of private (nonfarm) payroll employment; (5) the log of (real) personal consumption expenditures (PCE); (6) the log of the PCE price deflator; (7) the effective nominal federal funds rate; (8) the nominal 10-year Treasury yield; and (9) the value-weighted total stock market (log) return. We impose a Minnesota prior on the reduced-form VAR parameters by using dummy observations (see Del Negro and Schorfheide, 2011) and select the hyper-parameters that govern their prior distributions and the VAR lag length p by maximizing the marginal data density; to perform this optimization, we use the version of the CMA-ES evolutionary algorithm proposed by Andreasen (2010). The resulting specification, which includes a constant, is estimated over the 1975:M1–2012:M12 period using twelve lags of the endogenous variables.¹⁴

5 Main Results

This section presents our main results. First, we discuss the responses of macroeconomic variables to the identified financial and uncertainty shocks and the associated forecast error variance decompositions. In presenting these (and other similar) results, we adopt the following expositional scheme, which is best viewed in color. Specifically, the results based on the baseline identification of financial and uncertainty shocks use an orange-based color motif, with solid lines denoting implications of financial (EBP) shocks and dotted lines denoting implications of uncertainty (RVOL) shocks. The results based on the alternative identification of these two shocks use a green-based color motif, with solid lines denoting implications of uncertainty shocks and dotted lines denoting implications of financial shocks. In other words, the solid (dotted) lines are always referencing the implications of a shock that is identified first (second) in both identification schemes.

Our next set of results examines the historical contributions of financial and uncertainty shocks

 $^{^{14}}$ We use the first two years of the sample (1973:M1–1974:M12) as a training sample for the Minnesota prior. All the results reported in the paper are based on 10,000 draws from the posterior distribution of the structural parameters, where the first 2,000 draws were used as a burn-in period (increasing the number of draws had no effect on the reported results).

to U.S. economic fluctuations since the mid-1970s. Lastly, we corroborate our interpretation of the identified innovations as financial and uncertainty shocks by examining their correlations with other widely studied macroeconomic disturbances.

5.1 Macroeconomic Implications of Financial and Uncertainty Shocks

The solid lines in Figure 3 show the median impulse responses of the nine endogenous variables to a one standard deviation shock to the EBP under the baseline identification scheme, while the shaded bands represent the corresponding 90-percent pointwise credible bands. Overall, such an unanticipated jump in the EBP—only about 25 basis points—elicits an immediate increase in the RVOL measure of uncertainty and induces a significant deterioration in broad macroeconomic conditions. The fact that RVOL increases immediately in response to a rise in the EBP suggests that fluctuations in uncertainty may arise endogenously in response to changes in broader financial conditions.

On the real side, industrial production, an especially cyclically sensitive measure of economic activity, responds very quickly to the tightening of financial conditions and bottoms out a full percent below the trend about 14 months after the shock. The deterioration in labor market condition, while substantial, is a bit more gradual, with payroll employment reaching a trough about two years after the initial financial disruption. The ensuing labor market weakness is associated with a significant pullback in household spending, and the emergence of substantial and persistent economic slack induces a notable downward pressure on consumer prices. The prolonged slump engendered by an adverse financial shock is consistent with the growing empirical evidence, which shows that recoveries after "financial" recessions tend to be considerably more sluggish compared with "normal" recessions (see Cerra and Saxena, 2008; Reinhart and Rogoff, 2009; Bordo and Haubrich, 2010; Jorda et al., 2013).

On the financial side, the tightening of credit conditions and a rise in uncertainty lead to a sharp and immediate drop in the broad stock market, implying a substantial loss of wealth for the household sector that further saps consumer spending. The combination of growing economic slack and appreciable disinflation in the wake of the financial shock elicits a significant easing of monetary policy, as evidenced by the decline in the federal funds rate. And although longer-term interest rates decline noticeably in anticipation of such policy easing, the financial shock leaves stock prices permanently lower.

Figure 4 traces out the effects of a financial shock under the alternative identification scheme. In this case, an unanticipated increase in the EBP—again roughly about 25 basis points—has no meaningful economic or statistical effect on the RVOL measure of uncertainty. Nevertheless, the shock retains its hallmark features of a significant financial disruption. Factory output, employment, and consumer spending all register substantial and persistent decline in response to an adverse financial shock, and the downturn in economic activity is amplified importantly by the substantial drop in the stock market. In fact, the magnitude and shape of the responses of all macroeconomic variables to the EBP shock under the alternative identification scheme are very similar to those



Figure 3: Macroeconomic Impact of a Financial Shock (Baseline Identification Scheme)

NOTE: The solid line in each panel depicts the median impulse response of the specified variable to a 1 standard deviation shock to the EBP identified using the baseline identification scheme (see the text for details). Shaded bands denote the 90-percent pointwise credible sets.

implied by the baseline identification scheme shown in Figure 3. In combination, these results indicate that swings in uncertainty may not be very informative for gauging the macroeconomic effects of financial disturbances; at the same time, however, strains in credit markets will—in addition to depressing economic growth—likely lead to heightened economic uncertainty.

The solid lines in Figure 5 show the amount of variation in the same endogenous variables explained by financial shocks identified according to the first scheme, whereas the dotted lines show the same information for shocks implied by the alternative identification scheme. In both



Figure 4: Macroeconomic Impact of a Financial Shock (Alternative Identification Scheme)

NOTE: The solid line in each panel depicts the median impulse response of the specified variable to a 1 standard deviation shock to the EBP identified using the alternative identification scheme (see the text for details). Shaded bands denote the 90-percent pointwise credible sets.

cases, the identification of financial shocks involved selecting orthogonalized innovations in the EBP that maximized the response of this indicator of the tightness of financial conditions over the first six months after the impact of the shock. However, under both approaches, the results also imply that the identified financial shocks account for the bulk of the variation in the EBP at business cycle frequencies, a result consistent with the notion that exogenous changes in financial conditions are due primarily to financial disturbances. Consistent with the idea that fluctuations in uncertainty are partly a reflection of changes in the underlying financial conditions, innovations



Figure 5: Forecast Error Variance Decomposition of a Financial Shock (Baseline vs. Alternative Identification Scheme)

NOTE: The solid (dotted) line in each panel depicts the median estimate of the portion of the forecast error variance of a specified variable attributable to a 1 standard deviation shock to the EBP under the baseline (alternative) identification scheme (see the text for details). Shaded bands denote the 90-percent pointwise credible sets corresponding to the baseline identification scheme.

in the EBP implied by the baseline identification scheme explain a sizable portion of the forecast error variance in the RVOL measure of uncertainty. Under the alternative identification scheme, by contrast, financial shocks are completely uninformative about the future swings in uncertainty, a result consistent with the impulse response shown in Figure 4.

According to this metric, financial shocks are a significant source macroeconomic fluctuations they are estimated to explain between 25 and 35 percent of the variation in industrial production and payroll employment, and about 10 percent of the variation in consumer spending. In



Figure 6: Macroeconomic Impact of an Uncertainty Shock (Alternative Identification)

NOTE: The solid line in each panel depicts the median impulse response of the specified variable to a 1 standard deviation shock to the RVOL uncertainty measure identified using the alternative identification scheme (see the text for details). Shaded bands denote the 90-percent pointwise credible sets.

addition, such disruptions in the credit-intermediation process explain a significant portion of the variation in the stock market and short-term nominal interest rates. Overall, these proportions are in line with the recent evidence on the importance of credit shocks for business cycle fluctuations (see Gilchrist et al., 2009; Helbling et al., 2011; Meeks, 2012; Bassett et al., 2014a; Peersman and Wagner, 2014).

The macroeconomic implications of uncertainty shocks are shown in Figures 6 and 7. As evidenced by the response of the EBP, an unanticipated increase in financial market volatility

of 1 standard deviation—about 5 percent—leads to an immediate tightening in credit market conditions under the alternative identification scheme (Figure 6). The uncertainty shock also has significant adverse consequences for economic activity: Factory output, payroll employment, and personal consumption expenditures all decline noticeably, though their respective declines in response to a positive uncertainty shock are somewhat less pronounced and persistent than in the case when the economy is hit by an adverse financial shock.

The qualitative similarity between uncertainty and financial shocks is also echoed in the behavior of the stock market, which drops sharply in response to a jump in uncertainty. As in the case of a financial disruption, share prices remain depressed for a considerable period of time, and the broadbased deterioration in economic conditions elicits as significant easing of monetary policy. The fact that the response of key macroeconomic variables to a financial shock is qualitatively so similar to that of an uncertainty shock underscores the difficulty of assigning distinct sign restrictions to the IRFs in order to jointly identify the two types of disturbances.

Figure 7 shows the implications of a positive uncertainty shock identified using the baseline scheme, in which the optimization problem that selects the financial shock is solved before the problem that identifies the orthogonalized volatility innovation. The sole effect of such an uncertainty shock, which is quite persistent, appears to be a short-lived decline in the stock market that has no consequences for broad financial conditions or real economic outcomes.¹⁵ These results indicate that if uncertainty shocks are to have real economic effects, they must exert a significant independent effect on financial market conditions. And even though our identification schemes do not impose any zero restrictions on contemporaneous responses, identifying financial shocks first effectively shuts down this transmission channel.

The limited economic significance of uncertainty shocks in the baseline identification is corroborated by the dotted lines in Figure 8, which depict the contribution of such shocks to the forecast error variance of the endogenous variables of the VAR system. While uncertainty shocks implied by the baseline identification scheme explain a significant majority of the variability in the RVOL measure of uncertainty, their contribution to the forecast error variance of other variables—with the exception of the stock market—is indistinguishable from zero. Even under the alternative identification scheme, the significance of uncertainty shocks is, judging by this metric, considerably smaller than than of financial shocks under either identification scheme.

In combination, these results are consistent with theoretical mechanisms that emphasize the presence of frictions in financial markets—and their effect on the effective supply on credit—as an important conduit through which fluctuations in uncertainty are propagated to the real economy (see Arellano et al., 2012; Christiano et al., 2014; Gilchrist et al., 2014b). In response to an unanticipated increase in uncertainty, distortions in financial markets induce a tightening of financial conditions—effectively reducing the supply of credit available to businesses and households—which

¹⁵The highly persistent response of realized volatility appears to be due to our parametrization of the Minnesota prior, rather than to some underlying economic mechanism. In fact, when we used a "weaker" parametrization of the prior, the response of RVOL converges to its mean within 12 months, while the responses of all other variables remain qualitatively and quantitatively very similar to those reported in Figure 7.



Figure 7: Macroeconomic Impact of an Uncertainty Shock (Baseline Identification Scheme)

NOTE: The solid line in each panel depicts the median impulse response of the specified variable to a 1 standard deviation shock to the RVOL uncertainty measure identified using the baseline identification scheme (see the text for details). Shaded bands denote the 90-percent pointwise credible sets.

leads to a decline in spending and production and a drop in the stock market. The results also indicate that financial disturbances have a large effect on economic activity that is largely independent of the level of uncertainty in the economy. And lastly, the above evidence supports the view that a significant portion of fluctuations in financial market volatility may be a reflection of changes in the underlying economic and financial conditions (see Bachmann et al., 2013).



Figure 8: Forecast Error Variance Decomposition of an Uncertainty Shock (Baseline vs. Alternative Identification Scheme)

NOTE: The solid (dotted) line in each panel depicts the median estimate of the portion of the forecast error variance of a specified variable attributable to a 1 standard deviation shock to the RVOL uncertainty measure under the baseline (alternative) identification scheme (see the text for details). Shaded bands denote the 90-percent pointwise credible sets corresponding to the alternative identification scheme.

5.2 Historical Significance of Financial and Uncertainty Shocks

To put the above results into a historical perspective, this section examines the role of financial and uncertainty shocks in economic fluctuations over roughly the past 40 years. That is, for each variable, we calculate the portion of the actual series that is attributable to the two types of shocks over the 1975–2012 period. The results of this exercise for the baseline identification scheme are shown in Figure 9, while those based on the alternative identification are shown in Figure 10.

Figure 9: Historical Variance Decomposition (Baseline Identification Scheme)



NOTE: The shaded regions in each panel depict the historical contributions of shocks to the EBP (red) and the RVOL uncertainty measure (green) to the specified variable; the two shocks are estimated using the baseline identification scheme. The actual series (solid lines) are expressed in deviations from their respective estimated means (see the text for details).

To conserve space, we report the historical variance decomposition only for the EBP, the RVOL measure of uncertainty, the growth in industrial production, and the stock market return. (The results for the remaining endogenous variables are collected in Appendix B.)

According to the historical decomposition implied by the baseline identification scheme, the ups and downs in economic activity over the past two decades were shaped importantly by shocks emanating from the financial sector. Importantly, these shocks account for virtually all of the movements in the EBP, a strong indication that this financial indicator accurately captures disruptions in the credit-intermediation process. Uncertainty shocks implied by the baseline identification scheme, by contrast, do not appear as a significant source of business cycle fluctuations. Moreover, the identified financial shocks account, on balance, for about as much of the variability in the RVOL measure of uncertainty as do the uncertainty shocks, a result consistent with the notion that heightened economic uncertainty is in large part an outcome of financial instability.

Figure 10: Historical Variance Decomposition (Alternative Identification Scheme)



NOTE: The shaded regions in each panel depict the historical contributions of shocks to the EBP (red) and the RVOL uncertainty measure (green) to the specified variable; the two shocks are estimated using the alternative identification scheme. The actual series (solid lines) are expressed in deviations from their respective estimated means (see the text for details).

Nevertheless, the economic significance of financial shocks has varied considerably over the past four decades. Such shocks played a distinctly secondary role in economic fluctuations during the first half of our sample, a finding consistent with the heavy regulation of financial institutions and markets during this period and the empirical evidence showing that the economic downturns of the 1970s were influenced significantly by the OPEC-induced increases in oil prices (see Hamilton, 1983, 2003), while the recessions of the early 1980s owed importantly to the tightening of monetary policy under the then-Fed Chairman Volcker, who was determined to fight inflation and reverse the rise in inflation expectations (see Lindsey et al., 2005).

Financial shocks, however, assumed new importance in the wake of financial deregulation and the associated financial deepening that took place during the second half of the 1980s and the early 1990s. According to our estimates, a pronounced tightening in credit market conditions occurred during periods surrounding the 2001 and 2007–09 cyclical downturns, and adverse credit supply shocks account for significant portions of the decline in industrial output and equity valuations during these two recessions. On the other hand, easy financial conditions—at least in retrospect—characterized much of the mid-1990s and mid-2000s, and economic activity and the stock market were buoyed substantially by expansionary credit supply shocks.

As shown in Figure 10, a qualitatively similar historical narrative emerges under the alternative identification of the two types of shocks. As before, historical changes in credit market conditions are driven primarily by financial shocks, although adverse uncertainty shocks contributed notably to the massive and widespread tightening in financial conditions experienced at the nadir of the 2008–09 financial crisis. And while financial disturbances become a less important source of fluctuations in economic uncertainty under the alternative identification scheme, these shocks remain a significant source of macroeconomic instability during the second half of our sample period. A meaningful economic significance of uncertainty shocks, by contrast, appears to be confined to the "Great Recession" and its immediate aftermath. In general, uncertainty shocks have contributed systematically to swings in equity prices over the past 40 years or so, but their impact on the real economy—insofar they have induced a change in broad financial conditions—appears to have been quite limited.

5.3 Validation of Financial and Uncertainty Shocks

The historical variance decomposition presented above clearly indicates that the identified financial shocks—and to a lesser extent uncertainty shocks—were an important driver of fluctuations in economic activity and swings in broad equity prices since the mid-1980s. At the same time however, a natural question that emerges from this analysis is whether these two types of shocks in fact represent distinct sources of cyclical fluctuations or whether they are simply emblematic of traditional origins of macroeconomic instability.

To examine this hypothesis more formally, we look at the correlations between the identified financial and uncertainty shocks and other widely cited economic disturbances, all of which are external to our VAR system. At monthly frequency, we consider two types of popular shocks: monetary policy and oil price shocks. To measure unanticipated changes in the stance of monetary policy, we rely on high-frequency financial market data. Our first monetary policy shock corresponds to the "target surprise" proposed by Kuttner (2001), which measures the unexpected change in the target federal funds rate associated with an FOMC announcement.¹⁶ In addition, we use changes in the (on-the-run) 2-year Treasury yield over a narrow window bracketing FOMC announcements as monetary policy surprises, which provide a more complete characterization of the unanticipated changes in the stance of policy, according to Gilchrist et al. (2014a).¹⁷

¹⁶Specifically, the unanticipated change in the funds rate is calculated as the change—with minor adjustments—in the current-month federal funds futures contract rate in a 30-minute window (10 minutes before to 20 minutes after) around the FOMC announcement; see Kuttner (2001) for details. These target surprises, as they are commonly referred to in the literature, have been used extensively to examine the effects of monetary policy on asset prices (see Gürkaynak et al., 2005; Bernanke and Kuttner, 2005).

¹⁷As emphasized by Gürkaynak et al. (2005), using solely the target surprises to characterize the unanticipated changes in the stance of monetary policy is incomplete because such an approach omits the effect of changes in the

We also consider two monthly measures of oil price shocks. The first measure is the widely used nonlinear transformation of the nominal price of crude oil—the so-called net oil price increase proposed by Hamilton (2003). The second measure corresponds to the oil supply shock due to Killian (2009), who employs a SVAR-based approach to identify the underlying demand and supply shocks in the global crude oil market.

At quarterly frequency, we concern ourselves with technology and fiscal shocks. The first set of technology shocks corresponds to shocks to labor productivity identified by Mertens and Ravn (2011a) using a SVAR-based approach, which are orthogonal to tax shocks derived from the narrative approach of Romer and Romer (2010). As an alternative proxy, we also use a simple quarterly growth rate in the utilization-adjusted total factor productivity (TFP), which attempts to adjust measured TFP for a range of non-technological factors that can drive a wedge between TFP and technology (see Basu et al., 2006).

On the fiscal front, we consider the surprise tax policy changes from Mertens and Ravn (2011b), which are based on the narrative approach of Romer and Romer (2010); the anticipated tax policy changes of Leeper et al. (2013); and the unanticipated changes in the expected present value of government spending in response to military events from Owyang et al. (2013). All told, our list of external shocks spans the space of disturbances commonly considered to be the most important drivers of aggregate fluctuations.

The pairwise correlations between these external shocks and our identified financial and uncertainty shocks are reported in Table 5. As evidenced by the entries in the table, there appears to be no systematic contemporaneous association between financial and uncertainty shocks and other typical macroeconomic disturbances under either identification scheme. Virtually all pairwise correlations are statistically indistinguishable from zero and all of them are very small in economic terms. These results provide strong corroborative evidence for our view that the identified financial and uncertainty shocks represent independent sources of economic disturbances, with the former category playing an especially salient role in business cycle fluctuations over the past two decades.

Additional corroborative evidence for this view can also be gleaned from within our framework. A distinct feature of financial shocks emphasized in the literature is their effect on credit flows. In particular, an adverse financial shock should, by limiting the availability of credit, lead to a decline in bank lending to businesses and households—in fact, this aspect of financial shocks is used frequently to identify credit supply shocks using sign restrictions. To see whether the identified financial shocks are consistent with this intuitive feature, we augment our benchmark VAR specification with the log of real core loans outstanding.¹⁸ We employ the same two sets of

future policy rates that are independent of the shock to the current target rate and which are closely associated with the FOMC statements that accompany changes in the target rate. As shown by Gilchrist et al. (2014a), however, the first-order effects of monetary policy actions can be summarized adequately by the intraday changes in the 2-year nominal Treasury yield bracketing FOMC announcements; see also Hanson and Stein (2012) and Gertler and Karadi (2013).

¹⁸Core loans are the sum of loans to households and businesses. Business loans include commercial and industrial (C&I) loans and business loans secured by commercial real estate; household loans include residential mortgages, credit card loans, and other consumer loans. All monthly series were obtained from the Federal Reserve's H.8 Statistical Release. The household and business loans outstanding were deflated by the PCE price deflator.

	Identification Scheme				
	Ba	seline	Alte	rnative	
	Financial Uncertainty		Financial	Uncertainty	
A. Monthly External Shocks					
FFR target surprises ^a	0.04	-0.07	0.05	0.03	
2-year Treasury yield surprises ^b	0.04	-0.10	0.05	-0.08	
Net oil price increases ^c	-0.01	0.01	0.02	0.01	
Oil supply shocks ^d	0.10^{*}	-0.03	0.07	-0.00	
B. Quarterly External Shocks					
Technology shocks ^e	-0.09	0.08	-0.08	0.00	
TFP shocks ^f	-0.06	0.04	-0.11	0.10	
Unanticipated tax shocks ^g	-0.06	-0.05	-0.01	-0.09	
Anticipated tax shocks ^h	0.08	0.06	0.04	0.01	
Defense spending shocks ⁱ	0.02	0.06	-0.01	0.07	

Table 5: Correlations Between Financial, Uncertainty and Other External Shocks

NOTE: The entries in the table denote the pairwise correlations between the specified external shock and the financial (EBP) and uncertainty (RVOL) shocks identified under the baseline and alternative identification schemes. Monthly financial and uncertainty shocks are estimated over the 1975:M1–2012:M12 period; the quarterly shock series are the quarterly averages of the corresponding monthly values (see the text for details). * p < .10.

^a Unanticipated changes in the target federal funds rate in the 30-minute window bracketing FOMC announcements; see Kuttner (2001) (1992:M2–2012:M12, T = 251).

^b Changes in the (on-the-run) 2-year Treasury yield in the 30-minute window bracketing FOMC announcements; see Gilchrist et al. (2014a) (1992:M2–2012:M12, T = 251).

^c Net oil price increase; see Hamilton (2003) (1975:M1–2012:M9, T = 453).

^d Crude oil supply shock from Killian (2009) (1975:M1–2007:M12, T = 396).

^e Technology shocks from Mertens and Ravn (2011a) (1975:Q1-2006:Q4, T = 128).

^f Log-difference in the utilization-adjusted TFP; see Basu et al. (2006) (1975:M1-2007:M12, T = 152).

^g Unanticipated tax shocks from Mertens and Ravn (2011b) (1975:Q1–2006:Q4, T = 128).

^h Anticipated tax changes from Leeper et al. (2013) (1975:Q1-2006:Q4, T = 128).

ⁱ Defense spending shocks from Owyang et al. (2013) (1975:Q1–2012:Q4, T = 152).

assumptions to jointly identify financial and uncertainty shocks—it is worthwhile to note that this approach imposes no restrictions on the response of bank lending to both shocks. This augmented VAR specification is estimated over the 1986:M1–2012:M12 period, reflecting the availability of bank lending data.

Figure 11 traces out the response of bank lending to the two types of shocks.¹⁹ According to the left panel, unanticipated increases in the EBP lead to economically large and statistically significant declines in bank lending under both identification schemes. However, compared with the response of economic activity, the contraction in bank lending occurs with a significant lag. The delayed response of bank lending to an adverse financial shock importantly reflects the fact that

 $^{^{19}}$ We do not show responses of other variables in order to conserve space. However, under both identification schemes, the responses of all other variables were qualitatively and quantitatively very similar to those reported in Figures 3, 4, 6, and 7.



Figure 11: Financial and Uncertainty Shocks and Bank Lending (Baseline vs. Alternative Identification Scheme)

NOTE: The solid (dotted) line in the left panel depicts the median impulse response of real core loans to a 1 standard deviation shock to the EBP identified using the baseline (alternative) identification scheme; shaded bands denote the 90-percent pointwise credible sets corresponding to the baseline identification scheme. The solid (dotted) line in the right panel depicts the median impulse response of real core loans to a 1 standard deviation shock to the RVOL measure of uncertainty identified using the alternative (baseline) identification scheme; shaded bands denote the 90-percent pointwise credible sets corresponding to the alternative identification scheme; shaded bands denote the 90-percent pointwise credible sets corresponding to the alternative identification scheme. Sample period: 1986:M1–2012:M12 (see the text for details).

firms and households are financing a significant portion of their expenditures by drawing on their existing lines of credit rather than by borrowing through newly issued loans. In the early phases of a financially induced recession, this mechanically boosts the amount of loans on the banks' balance sheets and accounts for the sluggish response of loans outstanding to credit supply shocks (see Bassett et al., 2014a,b).

The response of bank lending to uncertainty shocks (the right panel) is also consistent with our earlier results, which showed that uncertainty shocks had real economic consequences only insofar they led to a change in broad financial conditions. Consistent with this result, positive uncertainty innovations have an economically and statistically significant effect on bank lending effect only under the alternative identification scheme, in which they elicit a concomitant tightening of financial conditions. Under the baseline identification scheme, in contrast, bank lending—like economic activity—shows no response to a jump in the RVOL measure of uncertainty.

6 Robustness Analysis

In this section, we perform several robustness exercises. First, we examine the sensitivity of the results based on the benchmark VAR specification to different sample periods. And second, we analyze the robustness of our results to different measures of uncertainty.

6.1 Alternative Sample Periods

We analyze the robustness of the results reported in Section 5 across two subsamples: (1) the 1985:M1–2012:M12 period; and (2) the 1985:M1–2007:M12 period. The first subsample corresponds to a period characterized by a stable monetary policy regime and by significant deregulation of financial markets. Reflecting the extraordinary events of the 2008–09 financial crisis, the second subsample omits the Great Recession and its aftermath. The results of this exercise are presented in Figure 12. To conserve space, we report only the responses of the EBP, industrial production, and the stock market.²⁰ For comparison purposes, Figure 12 also includes the benchmark results, which are shown in their original color motifs (see Figures 3, 4, 6, and 7).

The comparison of the solid red and the dotted black lines in Panels (a) and (b) reveals that the effects of financial shocks on economic activity and stock prices during the 1985–2012 period are, if anything, somewhat larger than those based on the full sample period. This result holds across both identification schemes and is consistent with the historical variance decomposition shown in Figures 9 and 10, which showed that financial shocks were an especially significant source of macroeconomic instability during that period. As shown by the dashed blue lines, dropping the Great Recession and its aftermath from the 1985–2012 subsample attenuates the near-term response of the macroeconomy to financial shocks. However, the longer-run responses are qualitatively and quantitatively similar to those from the full sample, an indication that our main results are not driven solely by this extraordinary period of financial market turmoil.

The macroeconomic effects of uncertainty shocks appear to be less robust across the different subsamples. Under the alternative identification scheme (Panel (c)), positive innovations in the RVOL measure of uncertainty during the 1985–2012 period elicit a tightening of financial conditions that is a bit more pronounced compared with that estimated over the full sample. Nevertheless, the response of industrial production is very similar to that from the benchmark specification, while the response of stock prices is somewhat larger over the first 12 months after the shock. Limiting the sample to the 1985–2007 period, by contrast, leads to a more muted EBP response. As a result, the effect of an uncertainty shock on industrial production disappears, highlighting the role that changes in credit market conditions play in propagating financial volatility shocks. Under the baseline identification scheme (Panel (d)), uncertainty shocks have no economically meaningful effect on industrial output in either of the two subsamples, although they continue to affect stock prices at the near-term horizon.

6.2 Alternative Uncertainty Proxies

Thus far, the macroeconomic effects of uncertainty shocks were based exclusively on the RVOL measure of uncertainty. This section examines the robustness of our results to alternative uncertainty proxies. In performing this analysis, we replace the RVOL uncertainty measure in the VAR specification with a different uncertainty proxy and identify financial and uncertainty shocks using

²⁰Responses of other variables are available from the authors upon request.



Figure 12: Macroeconomic Implications of Uncertainty and Financial Shocks (Alternative Sample Periods)

NOTE: See the text and notes to Figures 3, 4, 6, and 7 for details.



Figure 13: Macroeconomic Implications of Uncertainty and Financial Shocks (Uncertainty Proxies Based on Financial Market Data)

NOTE: See the notes to Figures 3, 4, 6, and 7 for details.



Figure 14: Macroeconomic Implications of Uncertainty and Financial Shocks (Uncertainty Proxies Not Based on Financial Market Data)

NOTE: See the text and notes to Figures 3, 4, 6, and 7 for details.

 (d) Response of selected variables to an uncertainty shock – baseline identification scheme

 the same two identification schemes as in the benchmark specification. Note that the resulting VAR specifications are estimated over different sample periods, reflecting differences in data availability.

Figure 13 shows the results of this exercise for uncertainty proxies based on equity valuations, while Figure 14 depicts the same information for proxies not based on financial market data. As before, we report only the responses of the EBP, industrial production, and the stock market in order to economize on space.²¹. As shown in Panels (a) and (b) of the two figures, the effects of financial shocks are completely robust with respect to different uncertainty proxies. Under both identification schemes, unanticipated increases in the EBP lead to a pronounced and persistent decline in industrial production and a substantial permanent drop in the stock market. As evidenced by the responses from specifications that includes the VXO and BBD uncertainty indexes, adverse financial shocks are estimated to lead to a considerably more adverse macroeconomic outcomes during the post-1985 period, a result consistent with the subsample analysis reported in Section 6.1.

The extent to which uncertainty shocks shape macroeconomic dynamics is less clear, however. Uncertainty proxies based on financial market data (Figure 13) generate fairly similar macroeconomic implications: In all cases, unanticipated jumps in financial market volatility induce a notable tightening of financial conditions under the alternative identification scheme (Panel (c)). It is worth noting that the response of the EBP to an IVOL uncertainty shock is somewhat more muted, reflecting the fact that this measure of idiosyncratic equity volatility is purged of forecastable variation in equity returns and thus is less likely to be contaminated by contractual and informational frictions that plague financial markets. As a result, the effects of IVOL uncertainty shocks on economic activity and stock prices are notably more attenuated compared with those of the RVOL shocks.²² Predictably, the macroeconomic effects of the IVOL and VXO uncertainty shocks under the baseline identification scheme (Panel (d)) are very similar to those from the benchmark specification: In both cases, increases in financial market volatility have no effect on the EBP or industrial production—they only induce a short-lived decline the stock market.

A similar picture emerges when we consider uncertainty shocks not based on financial market data (Figure 14). As shown in Panels (c) and (d), the BBD and BES uncertainty innovations have meaningful macroeconomic effects only under the alternative identification scheme, in which they induce a change in financial conditions; in fact, the response of both industrial output and the stock market to a BBD uncertainty shock is marginal even in that case, a result consistent with the poor predictive ability of this uncertainty proxy documented in Section 3.3. By contrast, the impact of financial shocks—under both identification schemes—stays the same, irrespective of the uncertainty measure included in the VAR system.²³

²¹Responses of other variables are again available from the authors upon request. For comparison purposes, both figures also includes the benchmark results, which are again shown in their original color motifs (see Figures 3, 4, 6, and 7)

²²The differences in the IRFs are not due to different sample periods because in this instance, the two VAR specifications are estimated over the same sample period.

 $^{^{23}}$ The noticeably more pronounced effect of the EBP shock on stock prices implied by the specification that uses the SCOT uncertainty proxy is a reflection of the 1991:M1–2012:M12 sample period, which is heavily influenced by the Great Recession.

7 Conclusion

This paper employs the penalty function approach to jointly identify shocks behind changes in financial conditions and economic uncertainty and to trace out the impact of these two types of disturbances on the economy. The two structural innovations are identified using a criterion that each shock should maximize the impulse response of its respective target variable over a pre-specified horizon, an approach that allows us to distinguish between shocks that have otherwise very similar qualitative effects on the economy. Intuitively, we assume that the unanticipated changes in financial conditions are due primarily to financial shocks, whereas fluctuations in economic uncertainty are driven mainly by uncertainty shocks.

We implement this approach in the context of a standard monetary VAR, augmented with the excess bond premium—an indicator of the tightness of financial conditions—and various proxies for macroeconomic uncertainty. The results indicate that financial shocks have a significant effect on the economy: An unanticipated increase in the EBP leads to a modest but persistent rise in uncertainty, a significant decline in economic activity, and a substantial drop in the stock market, dynamics consistent with our maintained assumption that the EBP shocks are capturing disruptions in the credit-intermediation process. Such credit supply shocks also account for a significant portion of the variation in economic activity and broad equity valuations since the mid-1980s.

The results also show that uncertainty shocks have a significant macroeconomic impact in situations where they elicit a meaningful response in the EBP—that is, a tightening of financial conditions; otherwise, the impact of uncertainty shocks on economic activity is substantially attenuated. The fact that uncertainty increases upon impact in response to an adverse financial shock also raises the possibility that fluctuations in uncertainty are influenced importantly by changes in financial conditions, suggesting that heightened economic uncertainty may be a general symptom of financial market disruptions.

References

- ADRIAN, T., E. MOENCH, AND H. S. SHIN (2010a): "Financial Intermediation, Asset Prices, and Macroeconomic Dynamics," Federal Reserve Bank of New York Staff Report No. 422.
- —— (2010b): "Macro Risk Premium and Intermediary Balance Sheet Quantities," IMF Economic Review, 58, 179–207.
- ADRIAN, T. AND H. S. SHIN (2010): "Liquidity and Leverage," Journal of Financial Intermediation, 19, 418–437.
- ANDREASEN, M. M. (2010): "How to Maximize the Likelihood Function of a DSGE Model," Computational Economics, 35, 127–154.
- ANG, A. AND G. BEKAERT (2007): "Stock Return Predictability: Is It There?" Review of Financial Studies, 20, 651–707.

- ARELLANO, C., Y. BAI, AND P. J. KEHOE (2012): "Financial Markets and Fluctuations in Volatility," Staff Report No. 466, Federal Reserve Bank of Minneapolis.
- ARIAS, J., J. RUBIO-RAMIREZ, AND D. WAGGONER (2013): "Inference Based on SVARs Identified with Sign and Zero Restrictions," Working Paper, Dept. of Economics, Duke University.
- BACHMANN, R., S. ELSTNER, AND E. SIMS (2013): "Uncertainty and Economic Activity: Evidence From Business Survey Data," *American Economic Journal: Macroeconomics*, 5, 217–249.
- BACHMANN, R. AND G. MOSCARINI (2012): "Business Cycles and Endogenous Uncertainty," Working Paper, Dept. of Economics Yale University.
- BAKER, S. R., N. BLOOM, AND S. J. DAVIS (2013): "Measuring Economic Policy Uncertainty," Chicago Booth Research Paper No. 13-02.
- BARSKY, R. B. AND E. R. SIMS (2011): "News Shocks and Business Cycles," *Journal of Monetary Economics*, 58, 273–289.
- BARTELS, R. (1982): "The Rank Version of von Neumann's Ratio Test for Randomness," *Journal* of the American Statistical Association, 77, 40–46.
- BASSETT, W. F., M. B. CHOSAK, J. C. DRISCOLL, AND E. ZAKRAJŠEK (2014a): "Changes in Bank Lending Standards and the Macroeconomy," *Journal of Monetary Economics*, 62, 23–40.
- BASSETT, W. F., S. GILCHRIST, G. C. WEINBACH, AND E. ZAKRAJŠEK (2014b): "Improving Our Ability to Monitor Bank Lending," in *Risk Topography: Systemic Risk and Macro Modeling*, ed. by M. K. Brunnermeier and A. Krishnammurthy, Chicago: The University of Chicago Press, 149–161.
- BASU, S., J. G. FERNALD, AND M. S. KIMBALL (2006): "Are Technology Improvements Contractionary?" *American Economic Review*, 96, 1418–1448.
- BERNANKE, B. S. AND K. N. KUTTNER (2005): "What Explains the Stock Market's Reaction to Federal Reserve Policy?" *Journal of Finance*, 60, 1221–1257.
- BLEANEY, M., P. MIZEN, AND V. VELEANU (2012): "Credit Spreads as Predictors of Economic Activity in Eight European Countries," Working Paper, University of Nottingham.
- BLOOM, N. (2009): "The Impact of Uncertainty Shocks," Econometrica, 77, 623–685.
- BLOOM, N., M. FLOETOTTO, N. JAIMOVICH, I. SAPORTA-EKSTEN, AND S. TERRY (2012): "Really Uncertain Business Cycles," NBER Working Paper No. 18245.
- BOIVIN, J., M. GIANNONI, AND D. STEVANOVIĆ (2013): "Dynamic Effects of Credit Shocks in a Data-Rich Environment," Staff Report No. 2013, Federal Reserve Bank of New York.
- BORDO, M. D. AND J. G. HAUBRICH (2010): "Credit Crisis, Money, and Contractions: An Historical View," *Journal of Monetary Economics*, 57, 1–18.
- BRUNNERMEIER, M. K. AND Y. SANNIKOV (2014): "A Macroeconomic Model with a Financial Sector," *American Economic Review*, 104, 379–421.
- CALDARA, D. AND C. KAMPS (2012): "The Analytics of SVARs: A Unified Framework to Measure Fiscal Multipliers," Finance and Economics Discussion Series Paper No. 2012-20, Federal Reserve Board.

- CAMPBELL, J. Y., M. LETTAU, B. G. MALKIEL, AND Y. XU (2001): "Have Individual Stocks Become More Volatile? An Empirical Exploration of Idiosyncratic Risk," *Journal of Finance*, 56, 1–44.
- CARHART, M. M. (1997): "On Persistence in Mutual Fund Performance," *Journal of Finance*, 52, 57–82.
- CARRIÈRE-SWALLOW, Y. AND L. F. CESPEDES (2013): "The Impact of Uncertainty Shocks in Emerging Economies," Journal of International Economics, 90, 316–325.
- CERRA, V. AND S. C. SAXENA (2008): "Growth Dynamics: The Myth of Economic Recovery," American Economic Review, 98, 439–457.
- CESA-BIANCHI, A., M. H. PESARAN, AND A. REBUCCI (2014): "Uncertainty and Economic Activity: A Global Perspective," Working Paper, Dept. of Economics, University of Southern California.
- CHRISTIANO, L. J., R. MOTTO, AND M. ROSTAGNO (2014): "Risk Shocks," American Economic Review, 104, 27–65.
- D'AMICO, S. AND A. ORPHANIDES (2008): "Uncertainty and Disagreement in Economic Forecasting," Finance and Economics Discussion Series Paper No. 2008-56, Federal Reserve Board.
- DEL NEGRO, M. AND F. SCHORFHEIDE (2011): "Bayesian Macroeconometrics," in *Handbook of Bayesian Econometrics*, ed. by J. Geweke, G. Koop, and H. van Dijk, Oxford University Press, Chapter 7.
- EICKMEIER, S. AND T. NG (2011): "How Do Credit Supply Shocks Propagate Internationally? A GVAR Approach," Deutsche Bundesbank Discussion Paper No. 27/2011.
- ELLIOTT, G. AND U. K. MÜLLER (2006): "Efficient Tests for General Persistent Time Variation in Regression Coefficients," *Review of Economic Studies*, 73, 907–940.
- FAMA, E. F. AND K. R. FRENCH (1992): "The Cross-Section of Expected Stock Returns," Journal of Finance, 47, 427–465.
- FAUST, J. (1998): "The Robustness of Identified VAR Conclusions About Money," Carnegie-Rochester Conference Series on Public Policy, 49, 207–244.
- FAUST, J., S. GILCHRIST, J. H. WRIGHT, AND E. ZAKRAJŠEK (2013): "Credit Spreads as Predictors of Real-Time Economic Activity: A Bayesian Model-Averaging Approach," *Review of Economics and Statistics*, 95, 1501–1519.
- FOSTEL, A. AND J. GEANAKOPLOS (2012): "Why Does Bad News Increase Volatility and Decrease Leverage," *Journal of Economic Theory*, 147, 501–525.
- FRY, R. A. AND A. R. PAGAN (2011): "Sign Restrictions in Structural Vector Autoregressions: A Critical Review," Journal of Economic Literature, 49, 938–960.
- GAMBETTI, L. AND A. MUSSO (2012): "Loan Supply Shocks and the Business Cycle," ECB Working Paper No. 1469.
- GERTLER, M. AND P. KARADI (2013): "Monetary Policy Surprises, Credit Costs, and Economic Activity," Forthcoming in the American Economic Journal: Macroeconomics.

- GERTLER, M. AND C. S. LOWN (1999): "The Information in the High-Yield Bond Spread for the Business Cycle: Evidence and Some Implications," Oxford Review of Economic Policy, 15, 132–150.
- GILCHRIST, S., D. LÓPEZ-SALIDO, AND E. ZAKRAJŠEK (2014a): "Monetary Policy and Real Borrowing Costs at the Zero Lower Bound," Forthcoming in the *American Economic Journal: Macroeconomics*.
- GILCHRIST, S. AND B. MOJON (2014): "Credit Risk in the Euro Area," NBER Working Paper No. 20041.
- GILCHRIST, S., J. W. SIM, AND E. ZAKRAJŠEK (2014b): "Uncertainty, Financial Frictions, and Investment Dynamics," NBER Working Paper No. 20038.
- GILCHRIST, S., V. YANKOV, AND E. ZAKRAJŠEK (2009): "Credit Market Shocks and Economic Fluctuations: Evidence From Corporate Bond and Stock Markets," *Journal of Monetary Economics*, 56, 471–493.
- GILCHRIST, S. AND E. ZAKRAJŠEK (2012): "Credit Spreads and Business Cycle Fluctuations," American Economic Review, 102, 1692–1720.
- GÜRKAYNAK, R. S., B. SACK, AND E. SWANSON (2005): "Do Actions Speak Louder Than Words? The Response of Asset Prices to Monetary Policy Actions and Statements," *International Journal* of Central Banking, 1, 55–93.
- HAMILTON, J. D. (1983): "Oil and the Macroeconomy Since World War II," *Journal of Political Economy*, 91, 228–248.
- (2003): "What is an Oil Shock?" Journal of Econometrics, 113, 363–398.
- HANSON, S. G. AND J. C. STEIN (2012): "Monetary Policy and Long-Term Real Rates," Forthcoming in the *Journal of Financial Economics*.
- HE, Z. AND A. KRISHNAMURTHY (2013): "Intermediary Asset Pricing," *American Economic Review*, 103, 732–770.
- HELBLING, T., R. HUIDROM, A. M. KOSE, AND C. OTROK (2011): "Do Credit Shocks Matter? A Global Perspective," *European Economic Review*, 55, 340–353.
- HODRICK, R. J. (1992): "Dividend Yields and Expected Stock Returns: Alternative Procedures for Inference and Measurement," *Review of Financial Studies*, 5, 357–386.
- HRISTOV, N., O. HÜLSEWIG, AND T. WOLLMERSHÄUSER (2012): "Loan Supply Shocks During the Financial Crisis: Evidence for the Euro Area," *Journal of International Money and Finance*, 31, 569–592.
- JORDA, O., M. SCHULARICK, AND A. M. TAYLOR (2013): "When Credit Bites Back," Journal of Money, Credit, and Banking, 45, 3–28.
- JURADO, K., S. C. LUDVIGSON, AND S. NG (2013): "Measuring Uncertainty," Working Paper, Dept. of Economics, New York University.
- KILLIAN, L. (2009): "Not All Oil Price Shocks are Alike: Disentangling Demand and Supply Shocks in the Crude Oil Market," *American Economic Review*, 99, 1053–1069.

- KURMANN, A. AND C. OTROK (2013): "News Shocks and the Slope of the Term Structure of Interest Rates," *American Economic Review*, 103, 2612–2632.
- KUTTNER, K. N. (2001): "Monetary Policy Surprises and Interest Rates: Evidence From the Fed Funds Futures Market," *Journal of Monetary Economics*, 47, 523–544.
- LEDUC, S. AND Z. LIU (2014): "Uncertainty Shocks are Aggregate Demand Shocks," Working Paper No. 2012-10, Federal Reserve Bank of San Francisco.
- LEEPER, E. M., T. B. WALKER, AND S. S. YANG (2013): "Fiscal Foresight and Information Flows," *Econometrica*, 81, 1115–1145.
- LINDSEY, D. E., A. ORPHANIDES, AND R. H. RASCHE (2005): "The Reform of October 1979: How It Happened and Why," *Review, Federal Reserve Bank of St. Louis*, 87, 187–235.
- MEEKS, R. (2012): "Do Credit Market Shocks Drive Output Fluctuations? Evidence From Corporate Spreads and Defaults," *Journal of Economic Dynamics and Control*, 36, 568–584.
- MERTENS, K. AND M. O. RAVN (2011a): "Technology-Hours Redux: Tax Changes and the Measurement of Technology Shocks," in *NBER International Seminar on Macroeconomics*, Chicago: The University of Chicago Press, vol. 7, 41–76.
 - (2011b): "Understanding the Aggregate Effects on Anticipated and Unanticipated Tax Policy Shocks," *Review of Economic Dynamics*, 14, 27–54.
- MOUNTFORD, A. AND H. UHLIG (2009): "What Are the Effects of Fiscal Policy Shocks?" Journal of Applied Econometrics, 24, 960–992.
- MUELLER, P. (2009): "Credit Spreads and Real Activity," Working Paper, London School of Economics.
- MUMTAZ, H. AND P. SURICO (2013): "Policy Uncertainty and Aggregate Fluctuations," CEPR Discussion Paper No. 9694.
- OWYANG, M. T., V. A. RAMEY, AND S. ZUBAIRY (2013): "Are Government Spending Multipliers Greater During Periods of Slack: Evidence from Twentieth-Century Historical Data," *American Economic Review*, 103, 129–134.
- PEERSMAN, G. AND W. WAGNER (2014): "Shocks to Bank Lending, Risk-Taking, Securitization, and Their Role for U.S. Business Cycle Fluctuations," Discussion Paper No. 2014-019, Tilburg University, Center for Economic Research.
- POPESCU, A. AND F. SMETS (2010): "Uncertainty, Risk-Taking, and the Business Cycle in Germany," *CESifo Economic Studies*, 56, 596–626.
- PRIETO, E., S. EICKMEIER, AND M. MARCELLINO (2013): "Time Variation in Macro-Financial Linkages," CEPR Discussion Paper No. 9436.
- REINHART, C. M. AND K. S. ROGOFF (2009): "The Aftermath of Financial Crises," *American Economic Review*, 99, 466–472.
- ROMER, C. D. AND D. H. ROMER (2010): "The Macroeconomic Effects of Tax Changes: Estimates Based on a New Measure of Fiscal Shocks," *American Economic Review*, 100, 763–801.

- ROUSSEEUW, P. J. AND C. CROUX (1993): "Alternatives to Median Absolute Deviation," *Journal* of the American Statistical Association, 88, 1273–1283.
- SCOTTI, C. (2013): "Surprise and Uncertainty Indexes: Real-Time Aggregation of Real-Activity Macro Surprises," International Finance Discussion Papers No. 1093, Federal Reserve Board.
- STOCK, J. H. AND M. W. WATSON (2012): "Disentagling the Channels of the 2007–09 Recession," Brookings Papers on Economic Activity, 44, 81–135.
- UHLIG, H. (2003): "What Moves Real GNP?" Working Paper, Humboldt University Berlin.
- (2005): "What Are the Effects of Monetary Policy on Output? Results From an Agnostic Identification Procedure," *Journal of Monetary Economics*, 52, 381–419.
- VAN NIEUWERBURGH, S. AND L. L. VELDKAMP (2006): "Learning Asymmetries in Real Business Cycles," *Journal of Monetary Economics*, 53, 753–772.

Appendices

A Uncertainty Measures

This appendix provides a brief description of the six uncertainty proxies used in the analysis.

RVOL measure of uncertainty. Sample period: 1973:M1–2012:M12. The realized equity volatility, our benchmark uncertainty proxy, is calculated as the (annualized) standard deviation of the daily value-weighted total market (log) return from the Center for Research in Security Prices (CRSP) data base. To mitigate the effects of large daily swings in equity prices—many of which occurred during the 2007–08 financial crisis—we use the robust estimator of scale proposed by Rousseeuw and Croux (1993) to calculate the monthly standard deviation of daily stock returns.

IVOL measure of uncertainty. Sample period: 1973:M1–2012:M12. This uncertainty proxy is a monthly version—with minor modifications—of the quarterly measure proposed by Gilchrist et al. (2014b). First, we extracted daily stock returns for all U.S. nonfinancial corporations with at least 500 trading days of data. This selection criterion yielded a panel of 14,706 firms over the period from October 1, 1972 to December 31, 2012. To ensure that our results were not driven by a small number of extreme observations, we eliminated all firm/day observations with a daily absolute return in excess of 50 percent.

The estimate of uncertainty is based on the following three-step procedure. First, we remove the forecastable variation in daily excess returns using the standard (linear) factor model:

$$(R_{it_d} - r_{t_d}^f) = \alpha_i + \beta'_i \mathbf{f}_{t_d} + u_{it_d}, \tag{A-1}$$

where *i* indexes firms and t_d , $d = 1, ..., D_t$, indexes trading days in month *t*. In equation (A-1), R_{it_d} denotes the (total) daily return of firm *i*, $r_{t_d}^f$ is the risk-free rate, and \mathbf{f}_{t_d} is a vector of observable risk factors. In implementing the first step, we employ a 4-factor model—namely, the Fama and French (1992) 3-factor model, augmented with the momentum risk factor proposed by Carhart (1997).

In the second step, we use the robust scale estimator of Rousseeuw and Croux (1993) to calculate the monthly firm-specific standard deviation of the daily idiosyncratic returns—that is, the OLS residuals \hat{u}_{it_d} from equation (A-1). Denoted by σ_{it} , this provides us with an estimate of time-varying equity volatility for firm *i*, a measure that is purged of the forecastable variation in expected returns and thus is less likely to reflect the countercyclical nature of contractual and informational frictions.

In the third step, we assume that the firm-specific measure of uncertainty σ_{it} follows an autoregressive process of the form:

$$\log \sigma_{it} = \gamma_i + \delta_i t + \sum_{k=1}^{3} \rho_k \log \sigma_{i,t-k} + v_t + \epsilon_{it}, \qquad (A-2)$$

where γ_i denotes a firm fixed effect intended to control for the cross-sectional heterogeneity in σ_{it} , while the firm-specific term $\delta_i t$ captures secular trends in the idiosyncratic risk of publicly traded U.S. nonfinancial firms documented by Campbell et al. (2001).

The IVOL uncertainty proxy corresponds to the sequence of estimated time fixed effects \hat{v}_t , $t = 1, \ldots, T$, which captures shocks to idiosyncratic volatility that are common to all firms. As emphasized by Gilchrist et al. (2014b), the presence of the common variation in the volatility of idiosyncratic equity returns is essential because if fluctuations in idiosyncratic volatility were themselves entirely idiosyncratic, the macroeconomic impact of such uncertainty shocks should wash out in the aggregate.

VXO measure of uncertainty. Sample period: 1986:M1–2012:M12. The VXO uncertainty proxy corresponds to the option-implied volatility calculated from a hypothetical at the money S&P 100 option 30 days to expiration.

BBD measure of uncertainty. Sample period: 1985:M1–2012:M12. This index of policy-related economic uncertainty is an amalgam of three components: (1) the frequency of newspaper references to policy-related economic uncertainty; (2) the number and revenue impact of federal tax code provisions set to expire in future years; and (3) the disagreement among professional forecasters on future government purchases and future inflation.

BES measure of uncertainty. Sample period: 1973:M1–2011:M12. To measure economic uncertainty, Bachmann et al. (2013) construct a measure of forecast dispersion using the Philadelphia Fed's Business Outlook Survey. This monthly survey of manufacturing firms contains qualitative information on the current state of firms' business conditions and their expectations of future business conditions. Bachmann et al. (2013) focus on two questions:

- 1. *General Business Conditions*: What is your evaluation of the level of general business activity six months from now versus [current month]? *Answers*: decrease; no change; increase.
- 2. Company Business Indicators: Shipments six months from now versus [current month]? Answers: decrease; no change; increase.

The qualitative survey responses to both questions are coded into three discrete numerical categories: -1 = decrease; 0 = no change; and 1 = increase.

For each question, Bachmann et al. (2013) define Frac_t^+ as the (unweighted) proportion of firms that responded with "increase" at time t and Frac_t^- as the (unweighted) proportion of firms that responded with "decrease" at time t. The cross-sectional forecast dispersion for any of the two questions is then computed according to

$$D_t = \sqrt{\operatorname{Frac}_t^+ + \operatorname{Frac}_t^- - \left(\operatorname{Frac}_t^+ - \operatorname{Frac}_t^-\right)^2}.$$

The measure of time-varying business-level uncertainty used in this paper corresponds to the crosssectional forecast dispersion for the question pertaining to general business conditions.

SCOT measure of uncertainty. Sample period: 1991:M1–2012:M12. This real-time uncertainty index measures the degree of uncertainty about real economic activity among financial market participants. Specifically, Scotti (2013) first estimates a dynamic factor model on actual economic data releases to construct monthly business conditions indexes and compute the associated forecasting weights. An economic data "surprise" corresponds to the difference between the actual data release and its median expectation from Bloomberg. The uncertainty index is computed as the square root of the weighted average of the squared surprises, where the time-varying weights are the forecasting weights from the estimation of the dynamic factor model in the first stage.

Figure A-1 shows the the time paths of the three uncertainty proxies that rely on financial market data (RVOL, IVOL, and VXO), while Figure A-1 shows the time-series evolution of uncertainty measures not based on financial market data (BBD, BES, and SCOT).

Figure A-1: Uncertainty Measures (Proxies Based on Financial Market Data)



NOTE: The panels of the figure show the three different measures of economic uncertainty based on equity valuations (see the text for details). The shaded vertical bars denote the NBER-dated recessions.

Figure A-2: Uncertainty Measures (Proxies Not Based on Financial Market Data)



NOTE: The panels of the figure show the three different measures of economic uncertainty that are not based on equity valuations (see the text for details). The shaded vertical bars denote the NBER-dated recessions.

B Historical Variance Decomposition



Figure B-1: Historical Variance Decomposition (Baseline Identification Scheme)

NOTE: The panels of the figure depict the historical contributions of shocks to the EBP (red) and the RVOL uncertainty measure (green) to the specified variable; the two shocks are estimated using the baseline identification scheme. See the main text and notes to Figure 9 for details.



Figure B-2: Historical Variance Decomposition (Alternative Identification Scheme)

NOTE: The panels of the figure depict the historical contributions of shocks to the EBP (red) and the RVOL uncertainty measure (green) to the specified variable; the two shocks are estimated using the alternative identification scheme. See the main text and notes to Figure 10 for details.